ERF Policy Research Report

Oil, Economic Diversification and Development in the Arab World

Ibrahim A. Elbadawi Alan H. Gelb





ERF Policy Research Report

Oil, Economic Diversification and Development in the Arab World

Ibrahim A. Elbadawi Alan H. Gelb

PRR No. 35 December 2010



First published in 2010 by The Economic Research Forum (ERF) 7 Boulos Hanna Street Dokki, Cairo Egypt www.erf.org.eg

Copyright © The Economic Research Forum, 2010

All rights reserved. No part of this publication may be reproduced in any form or by any electronic or mechanical means, including information storage and retrieval systems, without permission in writing from the publisher.

The findings, interpretations and conclusions expressed in this publication are entirely those of the author(s) and should not be attributed to the Economic Research Forum, members of its Board of Trustees, or its donors.

Preface

The Economic Research Forum (ERF) is undertaking a broad initiative to develop a critical mass of high quality research on a number of policy-relevant and high priority areas for the development of the Arab World. This initiative—The Research Initiative for Arab Development (RIAD)—is supported by a number of major international donors and will be implemented over many years.

This report aims at providing ERF with a sound basis for selecting and designing a research program over the next three to five years for one of the major themes of the initiative – oil management and economic diversification.

An earlier version of this report was presented at the workshop on "Natural Resources and Economic Diversification: Towards a Research Agenda for ERF", held in Cairo in November 2009. The draft was subsequently revised thanks to, among others, the valuable comments from the conference participants.

The report undertakes a review of the literature on the oil curse and economic diversification- two issues of vital relevance to the development of the Arab world. Not surprisingly agreeing with the with the near-consensus view that the curse is real but is not destiny, the report proposes a number of priorities for a future policy-relevant research agenda on the above two broad themes.

First, the macro-institutional issues for escaping the oil curse; and consequently the strategies for escaping it; or in other words, the plan for harnessing the oil resource for long-term development of the Arab world.

Second, the micro-institutional issues of economic diversification, where the report addresses some pivotal questions, such as why and how to diversify, and for what development objectives?

The authors would like to thank Carole Chartouni and Naotaka Sugawara for their able research support as well as the workshop participants for their valuable feedback and comments. Also, they stress that the views expressed are personal and should not be attributed to the Dubai Economic Council or the Center for Global Development.

Ibrahim Elbadawi Dubai Economic Council, UAE

Alan Gelb Center for Global Development, USA

Contents

Preface	iv
Contents	1
Chapter 1. Introduction	3
Chapter 2. Oil and Economic Diversification	7
Manufacturing and Export Concentration	7
Export Sophistication	
Proximity and Product Space	
Chapter 3. Oil Dependency	
Post-boom Growth Collapse	19
Resource Depletion and Negative "Genuine Savings"	
Chapter 4. Learning from the Resource Curse	
The Dutch Disease Perspective	
The Volatility Story	
The Governance Deficit and Rent-Seeking View	
Summing up	41
Chapter 5. Escaping the Oil Curse: The Macroeconomic Framework	
Fiscal Policy Rules	
The Political Economy of Natural Resource Funds (NRFs)	
Chapter 6. Issues for Future Research: Escaping the Oil Curse	
Understanding the Oil Curse	
The Macroeconomic Framework	
Chapter 7. Issues for Future Research: Economic Diversification	
A Three-Policy Country Framework	
Microeconomic and Institutional Research Issues	
Application of Product Space Concepts	
Prices, Factor Incentives, and Market Failures	
Comparative Case Studies	
Choice of Comparators	
Data	
Conclusions	

CHAPTER 1

Introduction

The Arab world is defined by oil.¹ In addition to their common cultural and historical heritage, the otherwise very diverse countries of this region are linked by their dependence on the hydrocarbon resource base: oil for short reference. Of the twenty-two member countries of the League of the Arab States, 11 are oil exporters. This group accounts for approximately 55 % of global oil reserves and 29% of natural gas reserves. Naturally, the hydrocarbon sector dominates these economies, where it contributes about 50% to GDP and 80% to government revenues (IMF, 2009).²

Moreover, given the linkages between the oil and non-oil Arab economies through the labor market, investment, and tourism, the latter group has also been substantially influenced by oil. The relatively populous non-oil Arab countries have depended on oil investment and aid from the oilrich, labor-importing Arab countries to finance development as well as to absorb part of their vast labor force as expatriate workers. Therefore, albeit in different ways, both oil and non-oil countries have been subject to the oil cycle and faced similar challenges.

Moreover, given the linkages between the oil and non-oil Arab economies through the labor market, investment, and tourism, the latter group has also been substantially influenced by oil. The relatively populous non-oil Arab countries have depended on oil investment and aid from the oilrich, labor-importing Arab countries to finance development as well as to absorb part of their vast labor force as expatriate workers. Therefore, albeit in different ways, both oil and non-oil countries have been subject to the oil cycle and faced similar challenges.

The received literature suggests that most Arab countries have experienced volatile, short-run growth and long-term stagnation. This has been linked to the failure of most countries to undertake medium-term, counter-cyclical macroeconomic policies. It is also linked to their failure to mediate conflicting interests during post oil booms due to their glaring lack of democracy, transparency, and accountability (Elbadawi, 2005a, 2005b). Related to this is that most, if not all, Arab countries continued to pursue old, state-led development strategies that, arguably, have outlived their effectiveness, as manifested in the massive unemployment crisis that afflicts the Arab world today (World Bank, 2004; Nabli, 2004) The disappointing growth and employment performance of the Arab world is mirrored in the low degree of diversification and sophistication of exports as well as the limited role of manufacturing in most Arab economies. Recent development experiences and the received literature suggest that countries that achieved high and sustained growth, by and large, have diversified economies and are endowed with good economic governance; and most, though not all, had large and dynamic manufacturing sectors (e.g. Imbs and Wacziarg, 2003; Hausmann, Hwang and Rodrik, 2006; UNIDO, 2009).

In this report, we agree with the view that oil is not destiny and that its ultimate impact on development hinges on the underlining institutional and policy environment. However, this report will also show that while oil resources have provided a huge opportunity to the Arab world to finance accelerated development; it has, nevertheless, complicated the development process in the region. This assessment is consistent with the consensus view in the received literature, which suggests that oil rents impede economic diversification and penalize manufacturing growth by generating Dutch Disease and extreme volatility. Oil rents also promote bad governance and complicate transition to transparent and accountable democratic rule. In addition, the oil sector tends to be located at the periphery of the product space, which makes it difficult for the economy to move into new and more sophisticated lines of products and services (Hausmann and Klinger, 2007).

Section one provides an overview of the role of oil in the Arab economy. Section two discusses the nexus of issues concerning economic diversification, manufacturing and export sophistication, and product space. This section will review the modest performance of the Arab world in these areas and highlight the likely negative effect of oil on this outcome. Section three discusses the evidence that oil economies, including those in the Arab world, have a tendency to experience frequent post-boom, growth deceleration and longterm stagnation, and in some cases absolute decline of output-in other words an "oil curse".

Section four undertakes a selective review of the literature on the oil curse, with the objective of identifying the most relevant issues for thinking through a strategy for escaping the oil curse or better, avoiding unstable growth and long-term economic stagnation. This will be addressed in section five, which covers some macro-institutional aspects of this strategy.

Sections six and seven deal with the key mandate of this work as an "approach" report designed to propose frontier areas for future research that are also of high policy relevance. Sections six and seven, respectively, cover the future research agenda with regard to macro-institutional solutions for escaping the oil curse and economic diversification of oil-dominated economies with special reference to the Arab region. Finally, section eight concludes this report.

1.1 The Role of Oil in the Arab Economy

Country evidence on oil reserves, shares in exports, and revenues show that Saudi Arabia alone accounts for 20% of global reserves and that oil

looms very large in the economies of these countries (Figures 1.1, 1.2, 1.3). However, before we further articulate the consequences of oil dependency for the region (section 2), we briefly discuss the differences between countries and groupings within the region. This is important to make explicit even as we emphasize the role of oil and its economic and political consequences that pervade the entire landscape of the Arab world. Indeed, the economies of the Arab countries are very diverse: in terms of size of economy and population, economic structures, level of development, geographic location, and type of governance and institutions. For example, in 2008 Qatar--one of the world's leading exporters of natural gas--had a population of 1.1 million and an average income per capita of \$86,000. This is almost 36 times the income per capita of Djibouti, the Arab country with a comparable population size (at 0.8 million); and almost 15 times the income per capita of Egypt, the most populous Arab country, with a population size of 75.2 million (IMF, 2009).

To highlight the economic diversity of the region more systematically, we follow the ERF (1998) and construct a typology of four Arab country groupings: oil-rich labor importing, which includes, in addition to Libya, the six countries of the Gulf Cooperation Council of Bahrain, Kuwait, Oman, Qatar, Saudi Arabia, and UAE; oil-rich, labor abundant (Algeria, Iraq, Sudan, and Yemen); Non-oil, middle income labor exporting (Egypt, Jordan, Lebanon, Morocco, Syria, and Tunisia);³ and, non-oil, low income (Djibouti, Mauritania).⁴

As can be seen from Table (1.1) the extent of the disparity across the four groups is clearly very stark. For example, in 2008 the oil-rich, labor importing group accounted for less than 14% of total Arab population but contributed more than 47% to total Arab GDP. The other oil-rich group is more populous but not as rich; with a population three times that of the oil-rich, labor importing countries, it accounted for less than half their contribution to the aggregate Arab GDP. In between the two groups are the non-oil, middle-income labor exporting countries, which constitute the largest grouping in terms of the number of countries and population; with more than 45% of the Arab population but only slightly more than 33% of the aggregate GDP.

Finally, in terms of average per capita incomes, the oil-rich, labor importing average is 3.4 times



Figure 1.1 Oil in the Arab World: Proven Reserves (percentage of world total: end of 2007)

Notes: 1. OTH means non-OPEC oil-exporting countries. 2. Oil-exporting countries are defined as those whose exports of oil and oil products (which are defined as Division 33 of SITC) exceed 20% of their merchandise exports.





Figure 1.3 Oil in the Arab World: Percentage of Total Revenue (average 2000-2008)



Table 1.1A Typology of Arab Economies

	Country Name	GDP Constant (USD bn)	Population	GDP per capita (USD Thou- sands)	% of Total Population to Overall Arab Popu- lation	% of GDP Constant to Total Arab GDP	% Ratio of Per Capita Income to Average Per Capita Income to the Arab World
Oil-Rich Labor import-							
ing (GCC + Libya)	Bahrain	27	1	27.00	0.31	1.07	343.20
	Kuwait	137.4	3.4	40.41	1.06	5.44	513.67
	Libya	88.1	6.2	14.21	1.93	3.49	180.62
	Oman	68.3	2.8	24.39	0.87	2.71	310.06
	Qatar	94.4	1.1	85.82	0.34	3.74	1090.83
	Saudi Arabia	592.9	24.9	23.81	7.76	23.49	302.66
	United Arab Emirates	185.3	4.8	38.60	1.50	7.34	490.70
	Total	1193.4	44.2	27.00	13.78	47.29	343.20
Oil-rich, Labor abundant	Algeria	233.5	34.8	6.71	10.85	9.25	85.29
	Iraq	105.8	30.4	3.48	9.48	4.19	44.24
	Sudan	88	38.1	2.31	11.88	3.49	29.36
	Yemen, Rep.	55.4	23	2.41	7.17	2.20	30.62
	Total	482.7	126.3	3.82	39.37	19.13	48.58
Non-oil middle income labor exporting	Egypt, Arab Rep.	443.4	75.2	5.90	23.44	17.57	74.95
	Morocco	137.1	31.4	4.37	9.79	5.43	55.50
	Jordan	32.4	5.9	5.49	1.84	1.28	69.80
	Lebanon	49.5	3.8	13.03	1.18	1.96	165.58
	Syrian Arab Republic	94.6	19.9	4.75	6.20	3.75	60.43
	Tunisia	82.6	10.3	8.02	3.21	3.27	101.93
	Total	839.6	146.5	5.73	45.67	33.27	72.85
Non-oil low-income	Djibouti	1.9	0.8	2.38	0.25	0.08	30.19
	Mauritania	6.2	3	2.07	0.94	0.25	26.27
	Total	8.1	3.8	2.13	1.18	0.32	27.09
Arab World		2523.8	320.8	7.87	100	100	100

the average income for the Arab world; compared to 0.73 and 0.49 times the Arab average income for the middle-income and oil-rich, labor abundant groups, respectively. As expected, the nonoil, low income countries accounted for only 0.27 times the Arab world average income per head. Interestingly, the data shows that the populous oil-rich countries of the Arab world are not that rich relative to the Arab mean income. On the other hand, Lebanon and Tunisia, two non-oil, middle income Arab countries, have done very well with the incomes per capita in the two countries, respectively, equaling three and two times the average for the populous oil-rich group.

Oil and Economic Diversification

iversification of new products correlates with economic development, especially sustained, export-oriented growth. Imbs and Wacziarg (2003) study the patterns of sectors concentrated across countries and time and find that income increases as economies become more diversified. Then economies start to specialize again at high levels of income.⁵ Very importantly from the perspective of industrial development, they also find that this process of diversification holds not only when economies transform from agriculture to manufacturing, but also within manufacturing.⁶ For example, Rodrik (2006a) argues that enhancing production capabilities for a large range of manufactured goods-including the production of new ones-is an integral part of development. Moreover, successful export-oriented strategies that lead to major economic transformations have been associated with significant economic diversification. For example, the empirical literature finds that non-traditional exports are characterized by higher income elasticity, less volatile terms of trade, and higher prospects of dynamic productivity gains (Elbadawi, 2002; Sekkat and Varoudakis, 1998). Also, following on the work of Haussman, Pritchett, and Rodrik (2005), researchers Johnson, Ostry, and Subramanian (2007) find that growth accelerations are associated with structural changes in manufacturing.

On the other hand, empirical research has found that countries that specialize in primary product exports tend to grow more slowly than economies with diversified export bases. Sachs and Warner

(1997) have shown that the 1970 share of primary exports in GDP is negatively correlated in growth regressions in a sample of 83 countries over the period 1965-1990. Sala-i-Martin (1997) has found a similar result for the 1970 share of primary products in total exports. In the special case of pointsource natural resources-those extracted from a narrow geographic or economic base such as oil and other minerals-the literature suggests that prudent and development-oriented management of oil and mineral booms has been the exception rather than the rule. Many countries inefficiently specialize in the resource sector and other nontradable activities that are likely to depend on it, which might lead to the collapse of output after the end of the boom-so-called "resource curse (e.g. Hausmann and Rigobon, 2001).

2.1 Manufacturing and Export Concentration

Lack of data precludes the calculation of concentration measures indexes, such as the Herfindahl-Hirschman index (HHI) for the major Arab oil exporters. However, as discussed in the introduction, the hydrocarbon sector looms very large in these economies, especially in the export sector. Moreover, the median level of the HHI concentration index for OPEC (at 66%) is the highest in the world (Figure 2.1).⁷ Even an emerging Arab oil economy, such as Sudan, has a relatively high concentration index (33%) for the period 1992-2005, though this index is likely to be much higher for the post 1998 period, when the country started to produce and export oil.⁸ On the other hand,



Figure 2.1 **Export Concentration Index (HHI):**

Notes: 1. Authors calculations from Elbadawi, Kaltani and Soto (2009) HHI data. 2. The median is taken for OPEC and only includes Venezuela, Nigeria, Ecuador and Angola due to data limitations.

some non-oil, middle-income Arab countries have achieved a substantial degree of export diversification. These include Lebanon, Morocco, and to a lesser extent, Jordan.

The concentration of exports from the Arab world is mirrored by the limited role of manufacturing in Middle Eastern economies, which are dominated by the Arab countries, especially when Turkey is excluded. The recent UNIDO Industrial Development Report (2009) contains extensive data on six components of industrial performance: manufacturing valued added (MVA) per capita; manufactured exports per capita; share of MVA in GDP; share of medium/high technology production in MVA; share of medium/high technology exports in manufactured exports; and share of manufactured exports in total exports. The evidence for 2005, the last year for which data is available, shows that for the first five indicators the Middle East and North Africa Region (MENA) uniformly underperformed Latin America and the Caribbean, and East Asia and the Pacific- the two regions at comparable levels of development;. MENA was only slightly better than the vastly poorer regions of South Asia and Sub-Saharan Africa. For the last indicator (the share of manufactured exports in total exports), which reflects the direct impact of dependency on the hydrocarbon resource exports in the region, MENA underperformed all of the four developing regions (Table 2.1).9

Recent literature draws a strong link between industrialization and the job-creation capacity of developing economies. For example, the abovementioned UNIDO report documents experiences of countries adopting labor-intensive, manufacturing-based development; The report finds that, overall, this development strategy creates jobs and that the job-creating capacity of strongly growing manufacturing sectors can be spectacular. Also, as industrialization proceeds not only employment expands, but also wages rise over time. Moreover, manufacturing is usually gender neutral with large number of women also being employed. Such experiences contrast sharply with the dominant role of the low-wage, informal sectors that have emerged as the main source of employment in non-oil Arab economies. Meanwhile, open or disguised unemployment among nationals in the GCC countries exposes the limits of their heavy dependence on the hydrocarbon and non-tradable sectors (e.g Assaad, 2002; Galal, 2002; World Bank, 2004; Elbadawi and Loayza, 2008). Therefore, viewing the unemployment crisis afflicting the Arab world, especially among educated youth, the region's failure to partake in the phenomenal global expansion of manufacturing exports over the last three decades has, by any measure, been a major development failure.

The received traditional development literature as well as the more recent Hausmann and Rodrik-led research on export sophistication, product space, and structural transformation advocate a major role for the state in addressing various types of market imperfections and coordination failures (see below). Unfortunately, however, the literature on Arab industrial development suggests that most of the region's governments have either totally abandoned this vital role, or instead, pursued ineffective traditional industrialization strategies that blunted these countries' ability to compete and their productivity growth (e.g. Sekkat, 2009).

2.2 Export Sophistication¹⁰

In addition to their failure to penetrate the global markets for low technology, labor-intensive manufacturing exports, the above evidence also reveals an equally disappointing performance for the Arab countries (MENA outside Turkey) with regard to the share of medium and high technology components of manufacturing. This,

Region/ Components	Manufacturing value added (MVA) per capita*	Manufactured exports per capita*	Share of MVA in GDP (percentage)	Share of manufactured exports in total exports	Share of medium/high- technology production in MVA (percentage)	Share of medi- um/high-tech- nology exports in manufac- tured exports (percentage)
Industrialized economies	4,771.0	5,428.2	16.8	85.7	75.2	66.1
Sub-Saharan Africa	63.6	81.2	10.9	62.0	41.3	32.4
excluding South Africa	29.2	38.5	7.6	54.9	17.5	13.3
South Asia	82.1	74.3	14.5	86.3	18.3	20.2
excluding India	79.6	51.9	15.9	84.6	5.2	8.0
Middle East and North Africa	398.1	474.7	12.5	31.7	33.3	27.9
excluding Turkey	381.4	367.1	12.1	22.7	19.2	20.0
Latin America and the Carib- bean	761.2	642.2	18.2	63.4	47.2	55.9
excluding Mexico	703.2	400.2	18.8	51.9	20.9	36.8
East Asia and the Pacific	582.3	885.6	29.5	91.9	97.5	64.1
excluding China	750.0	1 524 9	25.2	89.9	32.8	68.6

Table 2.1Industrial Performance in Developing Regions (2005)

Notes: 1. Authors' calculations from UNIDO Database, Industrial Development Report 2009; 2. *: MVA is in constant 2000 dollars.

we will argue, should be particularly worrisome for these countries. First, because they have little comparative advantage in basic low technology manufacturing, compared to the very low wages in Sub-Saharan Africa, South Asia, and especially China- the latter having essentially defined the frontier for labor-intensive manufacturing. Second, recent new research pioneered by Hausmann, Hwang, and Rodrik (2006) finds a strong association between the degree of export sophistication and subsequent economic growth. This evidence, argues Rodrik (2006a), suggests that "industrial upgrading is a leading indicator of economic performance" (p. 10) and that productivity levels associated with a country's exports are not fully captured by factors such as human capital or institutional quality.

The new index developed by these authors ranks traded goods in terms of their implied productivity. Thus for each country j, the index is given by:

$$EXPY_{j} = \sum_{l} \frac{x_{jl}}{X_{j}} PRODY_{l}$$
2.1

Where x_{jl} is the exports of product l by country j; $X_j = \sum_{j} x_{jl}$ is the total exports of country j; and PRODY₁ is the weighted sum of the per capita GDP of countries exporting a given product, where the weights reflect the revealed comparative advantage of each country in that product:

$$PRODY_{l} = \sum_{j} \frac{x_{jl} / X_{j}}{\sum_{j} (x_{jl} / X_{j})} Y_{j}$$
2.2

The higher this index is, the higher the content

of 'rich country products' in exports. This index is motivated by the view that "not all goods are alike in terms of their consequences for economic performance," and that specializing in some products will bring higher growth than specializing in others. In this setting, government policy has a potentially important positive role to play in shaping the production structure. Everything else being the same, countries that specialize in the types of goods that rich countries export are likely to grow faster than countries that specialize in other goods. Rich countries are those that have latched on to 'rich-country products,' while countries that continue to produce 'poor-country' goods remain poor.

Figure 2.2 presents estimates of EXPY in 2003 (the last year of the EXPY series calculated by Hausmann et al) for a few Arab countries and other comparators. Again, given its relatively high income, the above evidence makes clear that exports from the Arab world are characterized by relatively low export sophistication. For example, with only half of Saudi Arabia's income per capita, Malaysia's exports much more sophisticated products (of about 12000 units in the EXPY scale compared to less than 7000 for Saudi Arabia). Moreover, China provides even more spectacular contrasts where with an income per capita of only one fifth that of Saudi Arabia, the sophistication level of its exports were comparable to that of Ma-

Figure 2.2 Degree of Export Sophistication (EXPY: 2003)



Notes: 1. Authors' calculations from the data base on EXPY of Hausmann, Hwang and Rodrik (2006). 2. The data for Saudi Arabia is for the year 2002. 3. The data for Comoros is for the year 2000. 4. The median is taken for Non-Oil Middle-Income Labor Exporting countries which include Morocco, Jordan, Lebanon, and Syria. 5. The median is taken for OPEC countries which include Venezuela, Nigeria, Ecuador, Saudi Arabia, and Algeria.

laysia; and with an income approximately equal to the median for the non-oil middle-income Arab group, its EXPY index was almost twice the median score for the group. Finally, and despite their much higher incomes, the EXPY score for OPEC member countries was comparable to that of the non-oil middle-income Arab countries.

At this juncture, and not notwithstanding a few notable success stories,¹¹ we pose the question as to why this region's export performance has been so disappointing.

At first glance, it is natural to think that the dominance of the hydrocarbon sector must be an important explanatory factor. Controlling for income per capita and country fixed effects, the share of hydrocarbon fuels to total exports is positively associated with export concentration (HHI), while it has a strong negative impact on export sophistication (EXPY). Moreover, the residuals for large oil exporters fall above the line for HHI (EXPY),and below the line for EXPY, indicating that their performance could not be explained by their level of development and country fixed effects (Figures 2.3.a and 2.3.b).

Probing further, we review a recent paper by Elbadawi et al (2009) who estimate an empirical model of export performance (HHI, share of man-

Figure 2.3a





Notes: Based on the fixed-effects regression: Export concentration (HHI)= -0.0156***Per Capita GDP(PPP)+ 0.0004***Per Capita GDP2+0.0033*Share of Hydrocarbon fuels/total exports 0.185***Const; Where *** indicates significance at1%level. No. of observations= 1294; and No. of countries= 80.

Figure 2.3b

Export Sophistication (EXPY) and the Share of Hydrocarbon Fuel Exports.

Share of Hydrocarbon Fuels/Total Exports (%)



Notes: Based on the fixed-effects regression: Export Sophistication (EXPY)= 293.6556***Per Capita GDP(PPP)+-3.2801***Per Capita GDP2+-13.5386 Share of Hydrocarbon fuels/total exports+ 0.0912 Share of Hydrocarbon fuels/total exports2+ 5574.5618***Const; Where *** indicates significance at1% level. No. of observations= 1214; and No. of countries= 142.

ufacturing to total exports, EXPY) emphasizing the role of the real exchange rate as a determinant of profitability of tradable activities, especially non-traditional and new exports.¹² Conditional on other controls,¹³ they find strong association between RER overvaluation¹⁴ and the three indicators, with the former leading to larger export concentration, smaller share of manufacturing exports, and lower degree of export sophistication¹⁵ (Table 2.2). To underscore the significance of this finding, Elbadawi et al discuss specific country experiences in the context of a simple scatter plot showing EXPY against income per capita. They use this to highlight the exchange rate performance of countries that stand out as outliers relative to the level of EXPY predicted by their level of development (Figure 2.4). The scatter plot illustrates that there are many countries that are outliers, relative to the level predicted by a linear relationship between the two variables. Some of these countries have been able to export more sophisticated products than their development level would predict, while others export significantly below their predicted level. For example, China, India, and Indonesia, which maintained undervalued RER, belong to the former group; while Sudan belongs

Table 2.2

Export Diversification and Real Exchange Rate Misalignment. Cross-country Panel Data 1993-2004: Estimation Method: GMM-IV System Estimator

	De	pendent Variable	2
	Manufactur- ing Exports/ Merchandise Exports	Log of Herfindahl- Hirschman Index of Export Concentration	Log of EXPY
RER Misalignment	-0.0297**	0.0013**	-0.0007**
(% difference b/w log REER and its equilibrium)	0.0034	0.0005	0.0003
Standard Contro	l Variables		
Lagged Depen- dent Variable	0.9465**	0.7443**	0.7636**
	0.0041	0.0304	0.0317
Population	0.2227**	-0.0823**	0.0227**
(in logs)	0.0547	0.0118	0.0047
Real GDP per Capita	0.6782**	-0.1354**	0.0577**
(in logs)	0.1343	0.0232	0.0121
Land per Worker	-0.5469	-0.0219	0.0114
	0.9044	0.1681	0.0516
Landlock	0.1875	0.0717**	-0.0439**
	0.2103	0.0348	0.0133
Dummy for Ex- porters of Primary Products (not fuel)	-0.4297* 0.2359	0.0113 0.0371	-0.0626** 0.0162
Dummy for Ex-			
porters of Fuel	-2.0313**	0.4502**	-0.0230
	0.2879	0.0751	0.0148
Year Shifts	Y	Y	Y
No. Countries / No. Observations	66/551	62/670	55/385
Specification Tes	st (P-Values)		
(a) Sargan Test	n.a.	0.97	0.98
(b) Serial Cor- relation:	0	0	0.03
Second order	0.72	0.27	0.14

Notes: Numbers below coefficients are the corresponding robust standard errors. * (**) denotes statistical significance at the 10 (5) percent level.

Source: Table 7 of Elbadawi, Kaltani and Soto (2009).



Figure 2.4 Income Per Capita, EXPY, and Real Exchange Rate Misalignment

Notes: According to Elbadawi et al estimates: China, India and South Africa are countries that experienced and/or engineered extended RER undervaluation episodes, while the RER in Sudan has been overvalued for most of the post-1999 period. Source: Figure 12 of Elbadawi, Kaltani and Soto (2009).

to the latter and has been impacted by episodes of major RER overvaluation.¹⁶

China is a much-discussed country whose exports are described as special because they are much more sophisticated than its income level would indicate (Rodrik, 2006b)¹⁷ and also because of its exchange rate policy. Recent studies (e.g. Rodrik, 2007; and Johnson, Ostry and Subramanian, 2007) suggest that China's exchange rate has been undervalued (grossly so by Rodrik's estimates, nearly 50% in 2004, the last year of his analysis).¹⁸

It has been recently argued (most notably by Rodrik, 2007) that empirical findings like the one discussed above are, in fact, a reflection of a deeper causal effect that promises to open a new set of ideas for thinking about growth in which the RER takes center stage.¹⁹ According to Rodrik, countries that have managed to engineer an RER undervaluation appear to have resolved deep institutional constraints.²⁰ First, weak institutions create a wedge between private and social returns, which is different from simply having a low natural capacity for input. Second, to the extent that the tradable sectors may be more complex and entail more intensive transactions, the wedge between private and social returns may be more severe in tradable than non-tradable economic activities. This can lead to static misallocation of resources in favor of the latter and greater dynamic distortions in the former. When the tradable sector is more dynamic, as would be expected in many low-income, small economies, an increase in the relative prices of tradable to non-tradable commodities can improve static efficiency and enhance growth in a second-best fashion. Therefore, RER undervaluation can be the most feasible and effective approach for alleviating such institutional weaknesses. Another theoretical justification for engineering an RER undervaluation strategy is based on the view that tradable sectors (particularly new and non-traditional tradable sectors) are subject to a variety of market imperfections, such as information externalities (learning and cost-discovery externalities) and coordination externalities. These imperfections keep output and investment in tradable sectors at sub-optimal levels. Again, by raising the profitability of tradable sectors, an RER undervaluation can be an effective strategy in a second-best world. In particular, it can be an effective substitute to traditional industrial policy and all the well known limitations associated with it.

An important qualification, however, is provided by Eichengreen (2007), who argues that targeting certain sophisticated export activities by certain domestic policies, including those that promote RER undervaluation may merely play the role of a facilitating channel to permit the realization of certain favorable conditions. For example, he argues that to the extent that Chinese firms rely on their links to overseas Chinese or to their proximity to Japan and Korea, RER undervaluation or other domestic policies may not be enough for other countries that do not possess such an advantage.

Though the literature is still unfolding and will likely be subject to further refinements and qualifications, such as the issues raised by Eichengreen, there is, nevertheless, a compelling body of evidence to raise concerns about the growth prospects for the Arab world, given the dominance of oil and the prevailing policy environment, most notably with regard to the exchange rate policy pursued in most countries in the region (Elbadawi and Kamar, 2005). On the exchange rate issue, it appears that the median Arab country has not been able to avoid episodes of major overvaluation much less consistently target an undervalued RER as would be recommended by the recent literature. The exceptions were the cases of Tunisia and Morocco (representing the non-oil middle-income group), which have consistently pursued competitive real exchange rate policies. For the case of the oil rich, labor abundant (Figure 2.5.b) and non-oil middle income labor exporting Arab countries (Figure 2.5.c) the link between the RER undervaluation/ overvaluation and subsequent growth is very clear. In the former group extended RER overvaluation (RER misalignment index>0) episodes have been associated with growth deceleration/stagnation; while for the latter RER undervaluation has led to higher subsequent growth. However, for the other two Arab groups (oil rich, labor-importing, Figure 2.5.a; and non-oil low income, Figure 2.5.d) the relationship between RER and growth is less obvious. Not surprisingly most Arab countries could not match the stellar growth performances of countries like Chile and China, which have consistently pursued what seem to be real exchange rate-led growth strategies (Figure 2.5). Finally, going back to Table 2.2 (of Elbadawi et al regressions) we find that the dummy for exporters of fuel has a significant effect, influencing export performance in a similar fashion to that of the real exchange rate overvaluation. This suggests that there are factors beyond the basic controls and the real exchange rate that are specific to oil producing, exporting countries and are important for explaining the poor export performance of oil and other resource-rich economies.

2.3 Proximity and Product Space

Another important dimension of the recent research on export structure and sophistication that should be particularly relevant to an oil-rich economy is the concept of export clusters and product

Figure 2.5a Oil Rich, Labor Importing



Figure 2.5b Oil Rich, Labor Abundant





Figure 2.5f



Notes: 1. RERMIS (R) is the measure of the real exchange misalignment relative to a notional RER equilibrium level according to Rodrik (2007). 2. RERMIS (E) is the corresponding measure according to Elbadawi et al (2009). 3. RERMIS>0 implies RER overvaluation; and RERMIS<0 implies RER undervaluation. 4. A description of the calculation of the RERMIS is contained in the Appendix.

Figure 2.5d Non-Oil Low Income



Figure 2.5e China



space. The insight behind this research is based on two empirical regularities about product specialization and structural transformation across countries, established by Hausmann and Klinger (2007). They find that changes in the revealed comparative advantage of nations are governed by the pattern of what they call "relatedness" of products; and that this pattern tends to exhibit very strong heterogeneity: there are parts of the product space that are dense while others are sparse. This implies, they argue, that the structure of this product space governs the evolution of comparative advantage. Thus they develop a simple theoretical framework to formalize this point and to test the empirical relevance of their theory. In their review of the literature they claim that their findings are not accounted for by the received theory. For example, they argue that, controlling for factor endowments, the initial pattern of specialization has no independent effect in the Hechscher-Ohlin model (Leamer, 1987). Also they note that if we assume, as the Ricardian model argues, that technological differences across countries determine comparative advantage, the product mix will depend on the relative change of productivity across products. However, they once more claim that in the existing literature (such as the quality ladders or variety models: Grossman and Helpman, 1989; Aghion and Howitt, 1992) developing a new variety does not depend on the existing product mix. Hence, they argue, the structure of the product space has no importance

and cannot be a source of path dependence in the context of the received literature.

Against this backdrop, the two authors develop a simple product-specific model of human capital that also allows for a heterogeneous degree of substitutability across products. Fixing the output of each skilled worker to 1, goods are ordered in a line so that their prices increase with distance. Assuming that a worker needs to be trained to acquire the product-specific skills to be able to move from one product to another higher value added one, the additional revenues earned by the trained worker for moving from producing the current good i to another good j are given by:

$$\Delta P_{i,j} = f \delta_{i,j} \tag{2.3}$$

Where $\delta_{i,j}$ is the distance from good i to good j, equal to zero for i=j and greater than zero for i is not equal to j. Given the assumption that the degree of product substitutability declines with distance, the additional costs from moving from the current good i to good j is given by:

$$C(\boldsymbol{\delta}_{i,j}) = \frac{c\boldsymbol{\delta}_{i,j}^2}{2}$$
 2.4

The profit maximization associated with this simple problem, the optimal distance to jump and the maximum profit to be earned are, respectively, given by the following equations:

$$\underset{\delta_{i,j}}{Max} \prod = f \delta_{i,j} - \frac{c \delta_{i,j}^2}{2}$$
 2.5

$$\delta_{i,j}^* = \frac{f}{c}$$
 2.6

$$\Pi(\delta_{i,j}^*) = \frac{f^2}{2c}$$
 2.7

The distance profile of the profits is humped shaped (Figure 2.6). Remaining at the existing product i no extra profits are gained, but profits increase with distance until it reaches the maximum at δ^* , then it starts to decline until reaching zero at a distance equal to $\frac{2f}{c}$. This simple model shows that economies can stagnate if they specialize in products that are distant (in this model beyond $\frac{2f}{c}$) from other more dynamic or higher

value products. The implications of this model can be generalized to a multi-dimensional product space with n number of goods, represented by a matrix of (nxn) pair-wise distances. Also since the product space doesn't have to be a continuum, a break in this space larger than $\frac{2f}{c}$ will mean that an incremental jump will not be privately profitable, hence the economy will not experience incremental structural transformation. Therefore, such breaks, which these authors show are more common than not, would represent coordination failures when jumps that are not privately profitable are, nevertheless, socially optimal, since the newly created human capital has positive externality for future firms.

To examine the implications of this literature for oil-rich economies, we can probe further by discussing the empirical proxy for product proximity (the inverse of the distance concept). Mapping out the estimated measures of proximity into the product space allows analysis of its structure, including the location of the oil sector relative to other clusters (Hidalgo, Klinger, Barabasi and Hausmann, 2007). Moreover, the relationship between structural transformation and proximity or density of clusters can be empirically tested using this data (Hausmann and Klinger, 2007).

The empirical measure of the concept of proximity (relatedness), $\phi_{i,j}$, between two products i and j is given by the minimum of the pair-wise conditional probability of a country exporting a good given that it exports another:

 $\phi_{i,j} = \min\{P(RCAx_i \mid RCAx_j), P(RCAx_j \mid RCAx_i)\}$

Where RCA stands for revealed comparative

28

Figure 2.6 Distance Profile and Profits



Source: Figure 1 of Hausmann and Klinger (2007).

advantage

$$RCAc, i = \frac{x(c,i) / \sum_{i} x(c,i)}{\sum_{c} x(c,i) / \sum_{c,i} x(c,i)}$$
2.9

Which measures whether country c exports more of good i, as a share of its total exports, than the average country (RCA>1 not RCA<1). ²¹

Using the above two equations and four-digit level international trade data, Hausmann and associates calculate a matrix of revealed proximities between every pair of products. By mapping out the computed proximity measures into a "product space" they (Hidalgo, Klinger, Barabasi and Hausmann, 2007) show a heterogeneous space, composed of very dense clusters, where products are very closely connected. Meanwhile it is clear from the figure that oil and the majority of products fall at the periphery of the product space (Figure 2.7).²² On the other hand, unlike oil, which requires highly specific assets, manufactured products are found at the core of the product space, where the product clusters tend to be very dense. These authors, therefore, argue that the ability of countries to produce new products and undergo structural transformations depends on how many products are in close proximity to their current product mix. As noted by Hidalgo et al, in theory the extent of closeness between products, and hence the density of clusters, should depend on fundamental factors such as intensity of labor, land, and capital; the level of sophistication of the underlining technical, institutional, and human capital; or the logistics networks, inputs or outputs involved in the value chain.

Producing new products requires a combination of specific private and public input. If the new product is in close proximity to the current basket of goods, the private sector may be able to exploit the existing capabilities on its own the markets, physical and human assets, norms, and institutions that were set up for other preexisting activities (Hausmann and Rodrik 2006). To produce new products that are further away from the current product mix would necessitate certain capabilities that will not emerge on their own due to coordination problems. The government's intervention would be crucial in providing complementary input - rules, organization, infrastructure, labor training, and others - that are specific to subsets of activities. In South Korea and

Taiwan, they have taken the form of export subsidies; Singapore and Malaysia created an export processing zone; the Chinese government set up special economic zones. What is evident is the fact that the intervention has to fit the specifics of the context and hence is not fully knowable ex ante.

To recapitulate, the various findings from the classical literature on export concentration and the more recent research on export sophistication and product space all suggest that the dominance of the oil sector in the Arab world appear to have complicated the accelerated structural transformation and sustained growth through a variety of channels. The countries of the Arab region have, by and large, been less successful at diversifying their economies; breaking into the global market for manufacturing or producing and exporting higher value and more sophisticated exports commensurate with their levels of income. The oil sector tends to be located at the periphery of the product space, which places a huge demand on the public sector to develop the institutional capabilities to resolve the ensuing market failures.

However, the dominance of the oil sector might itself present a corrosive influence on institutions and governance in oil-rich countries. We turn to these issues in the next two sections.

Figure 2.7

Network Representation of the Product Space



Notes: 1. Source: Figure 1.B of Hidalgo, Klinger, Barabasi and Hausmann (2007). 2. Links are color coded with their proximity value. The sizes of the nodes are proportional to world trade, and their colors are chosen according to the classification introduced by Leamer (1984).

CHAPTER 3

Oil Dependency

"Projects of mining, instead of replacing capital employed in them, together with ordinary profits of stock, commonly absorb both capital and stock. They are the projects, therefore, to which of all others a prudent law-giver, who desired to increase the capital of his nation, would least choose to give any extraordinary encouragement..." (Adam Smith (1776)²³

s the above quote suggests, the concern about the resource curse at least dates back to the days of Adam Smith. However, sweeping statements about the perils of natural resource dependency for development, such as the above, are no longer widely accepted. There is now a near-consensus that a natural resource curse is real but is not destiny. On one hand, oil is a valuable resource that has been associated with high per capita incomes. Moreover, many, though not all, oil-rich countries have managed to successfully transform the rents from oil into substantial gains in terms of higher consumption and other aspects of social welfare for their citizens and investments in useful public goods for their economies (Sachs, 2007). A comparison of oilrich to non-oil developing countries makes clear that the former do in fact outperform the latter in terms of a wide range of economic and social indicators (Table 3.1). Also, as according to the overall development indicators, the major oil-rich Arab member countries of the GCC are obviously in a much bigger league than other countries in the GCC (Table 3.2).

Nevertheless, the oil curse is real and is mani-

fest in many important aspects of the development discourse of oil-rich countries. Oil income is intrinsically temporary because it is derived from non-renewable, depleting stocks. It is also unreliable because oil prices are highly volatile (Collier et al, 2009). Therefore, the consequences of failures to properly manage the volatility of oil incomes; or to effectively use it to accumulate large and sufficiently diverse stocks of tangible (e.g. infrastructure) and intangible (e.g. human and knowledge capital, good institutions) types of capital are likely to be extremely dire for oilrich countries.

In addition to the problems of limited economic diversification and low sophistication of exports that tend to afflict oil-rich countries, analyzed in the previous section, we briefly discuss below two more features common to oil-rich economies. These are extreme volatility, post-boom growth collapse, and inadequate renewable capital base, or in other words, very low or negative "genuine savings". Oil-rich economies affected by such problems are likely to be ripe for the oil curse or at least to experience long-term stagnation.

3.1 Post-boom Growth Collapse²⁴

The volatility of oil prices during the past three decades makes it possible to analyze countries' experiences during and in the aftermath of booms. Table 3.3 summarizes the experiences of oil and non-oil Arab countries and other non-Arab comparators during the first oil price boom of 1971–1980, the oil price collapse of 1981–1986, the intermittent period of 1987–1999, and finally

Economic and Social In	alcators in the	Arab World ar	na Comparators (20)	(2)					
	Arab Oil Rich, Labor Importing	Arab Oil-Rich, Labor Abundant	Arab Non-oil Middle- Income Labor Exporting	Arab Non-Oil Low-Income	Russian Federation	China	Malaysia	Indonesia	Sub- Saharan Africa
Energy production (barrels/ capita)**	235.11	6.15	2.52	NA	6.11	0.95	2.68	0.98	0.70
Life expectancy at birth, total (years)	76	63	73	64	68	73	74	71	52
Mortality rate, infant (per 1,000 live births)	11	55	24	75	13	19	10	25	89
GDP per capita, PPP (con- stant 2005 international \$)	28069	2218	4945	1820	14917	5511	13139	3674	1935
School enrollment, primary (% net)	06	75	89	63	NA	NA	97	95	72
Roads(KM/paved/1000 pop)	2.0	0.3	1.3	1.2	5.3	1.7	2.7	1.0	NA
Electricity production (kWh/capita)	14303	246	1676	NA	6974	2185	3506	597	NA

Table 3.1 Economic and Social Indicators in the Arab World and Comparators (2 Notes: 1. Authors' calculation from World Bank's WDI data base, 2009. 2. Data for the last available year is used. 3. **1 KT=0.00071380802249246 Millions of barrels.

20

OIL, ECONOMIC DIVERSIFICATION, AND DEVELOPMENT IN THE ARAB WORLD

the new price surge since 2000 until 2008. This vear witnessed the climax of the boom, where the price of oil reached a staggering \$140 pb around July before collapsing to less than \$40 during the second half of the year as a result of the current global economic crisis. The median oil country experience during the first boom was one of growth of GDP per capita above the non-oil exporting developing countries accompanied by high savings. However, growth in the median oil exporting country totally collapsed when fortunes were reversed. This post-boom period drastically distinguishes oil exporters from other developing countries. Savings, on the other hand, also experienced a large adjustment after the boom but still remained above those of non-oil developing countries. Even on this score, however, these savings might not be high enough compared with the optimum levels consistent with non-renewable resource base economies, such as the oil economies.²⁵ The median story thus shows that growth in oil-rich countries has been volatile and some have experienced negative growth rates after the end of the oil boom.

Taking a closer look at individual country experiences of oil exporters, three groups can be distinguished. One group of countries is made up of those that can be labelled success stories given their ability to continue growing even when fortunes have reversed (this would be in the 1980s period). Figure 3.1.a contains four panels depicting the income per capita in PPP terms for Indonesia and Norway but also for Botswana and Chile, which are not oil exporters but nevertheless mineral exporters and have managed to avoid the resource curse. Thus success stories exist for a variety of countries whose exports are primary products. The second group comprises those countries that epitomize the oil curse story. These include Nigeria as well as Saudi Arabia and Kuwait from the Middle East. They are depicted in Figure 3.1.b. Finally, there is another group of oil producing SSA countries that are relatively new entrants into the market, and for this reason they are yet to experience an oil price bust. Precisely because of this they can stand to learn from older oil-exporters in the other two categories. These countries are depicted in Figure 3.1.c; they are Chad and Sudan.

Perhaps the starkest contrast is provided by the comparison of Norway and Saudi Arabia. Per capita GDP (in PPP terms) in Norway rose precipitously from about US \$25,000 in 1980 to US

Table 3.2

Human Development Indicators in the Arab world and Comparators (2008)

	Country	HDI ranking
Arab Oil-Rich Labor Im-		
porting (GCC + Libya)	Bahrain	39
	Kuwait	31
	Libya	83
	Oman	56
	Qatar	33
	Saudi Arabia	59
	UAE	98
Arab Oil-rich, Labor		
abundant	Algeria	104
	Iraq	1001
	Sudan	150
	Yemen	140
Arab Non-oil Middle		
Income Labor Exporting	Egypt	123
	Morocco	130
	Jordan	96
	Lebanon	83
	Syria	150
	Tunisia	98
	West Bank Gaza	35
Arab Non-oil Low-Income	Comoros	139
	Djibouti	155
	Mauritania	154
Non-oil & Mineral non-oil		
Arab Countries	Botswana	125
	Chile	44
	Norway	1

Source: UNDP data base.

\$50,000 in 2008. Meanwhile, Saudi Arabia's per capita GDP declined from a peak of US \$35,000 in 1980 to slightly more than US\$ 15,000 in 1990 and only hit US \$20,000 in 2008. Therefore, while Norway's income rose steadily to surpass the OECD median, Saudi Arabia's income seems to have stagnated and could not fully recover from the early 1980s oil slump (Figure 3.2).

Table 3.3 Growth, Vola	ıtility and S	avings ac	ross Oil C	ycles									
			1974-1980			1981-1986			1987-1999			2000-2008	
Arab Group- ings	Countries	Growth	Volatility	Savings/ GDP	Growth	Volatility	Savings/ GDP	Growth	Volatil- ity	Savings/ GDP	Growth	Volatility	Savings/ GDP
Oil-rich, Labor Importing Countries (GCC+ 1 ihva)	Bahrain	ΝN	ΝN	55 00	70 4- 70 4-	1.36	47.02	۲4 4		69 22	3.67	0.35	40.91
	Kuwait	-7.05	1.62	61.51	-6.09	1.94	32.69	-0.10	105.69	13.90	4.56	1.29	41.74
	Libya	NA	NA	NA	NA	NA	NA	NA	NA	17.56	1.98	1.57	27.61
	Oman	3.88	2.58	48.33	7.59	0.68	41.16	1.13	3.33	26.08	3.31	0.61	37.67
	Qatar	NA	NA	NA	NA	NA	NA	NA	NA	42.55	09.0	10.28	67.01
	Saudi Arabia	3.89	2.24	60.08	-8.30	0.76	27.21	-0.34	12.23	24.44	1.59	1.48	44.55
	United Arab Emir- ates	-0.33	31.79	73.17	-11.19	0.68	53.44	-0.07	75.67	37.00	2.11	2.01	38.78
	median	1.77	2.41	60.08	-6.09	0.76	41.16	-0.07	12.23	26.08	2.11	1.48	40.91
Oil-rich, Labor Abundant	Algeria	2.91	0.96	39.42	0.83	2.42	34.61	-0.73	3.73	28.12	2.33	0.71	51.12
	Sudan	2.51	3.61	7.64	-1.50	3.77	4.76	2.56	1.90	8.22	5.31	0.38	16.21
	Yemen, Rep.	NA	NA	NA	NA	NA	NA	1.57	2.13	7.26	0.96	0.80	21.58
	median	2.71	2.28	23.53	-0.33	3.10	19.68	1.57	2.13	8.22	2.33	0.71	21.58
Non-oil Middle Income Labor Exporting	Egypt, Arab Rep.	6.25	0.66	14.13	3.30	0.77	14.91	2.10	0.70	14.76	2.94	0.60	15.16
	Morocco	3.13	0.86	13.58	1.83	2.66	15.69	1.11	5.47	18.15	3.53	0.63	23.41
	Jordan	11.75	0.64	-13.04	1.57	1.57	-10.95	-1.31	5.12	4.18	3.55	0.51	-8.41
	Lebanon	NA	NA	NA	NA	NA	NA	2.95	6.44	-23.34	2.88	1.08	-2.86
	Syrian Arab Re- public	7.33	1.15	14.41	-2.00	2.69	12.03	1.95	3.05	14.77	1.10	1.08	22.01
	Tunisia	4.16	0.41	25.20	0.56	6.13	23.06	2.71	0.91	22.06	3.86	0.38	22.01

Oil, Economic Diversification, and Development in the Arab World

22

\mathcal{C}	
<u></u>	
65	
e	
-	
b i	
<u> </u>	C

Continued													
			1974-1980			1981-1986			1987-1999			2000-2008	
Arab Group- ings	Countries	Growth	Volatility	Savings/ GDP	Growth	Volatility	Savings/ GDP	Growth	Volatil- ity	Savings/ GDP	Growth	Volatility	Savings/ GDP
	West Bank and Gaza	NA	NA	NA	NA	NA	NA	4.23	1.03	-19.58	-5.28	1.72	-28.48
	median	6.25	0.66	14.13	1.57	2.66	14.91	2.10	3.05	14.76	2.94	0.63	15.16
Non-oil Low- Income	Comoros	NA	NA	-10.08	1.21	1.34	-4.02	-0.98	4.14	-4.29	-0.25	6.79	-8.64
	Djibouti	NA	NA	NA	NA	NA	NA	-4.53	0.55	-6.42	1.04	1.51	5.78
	Mauritania	0.33	17.91	-5.49	-0.91	3.90	-0.54	-0.09	43.00	4.77	1.56	2.16	0.87
	median	0.33	17.91	-7.78	0.15	2.62	-2.28	-0.98	4.14	-4.29	1.04	2.16	0.87
Other	East Asia	2.18	1.00	31.71	2.34	0.40	31.25	2.50	0.72	32.39	3.12	0.43	29.01
	Latin America & Caribbean	2.81	0.51	23.07	-0.97	2.83	22.46	0.89	2.14	20.46	2.34	1.05	21.33
	Sub- Saharan Countries	0.50	4.33	23.69	-1.61	0.77	19.78	-0.46	3.63	16.22	2.26	0.57	16.60
	median	2.18	1.00	23.69	-0.97	0.77	22.46	0.89	0.72	20.46	2.34	0.57	21.33

Note: Computed from the World Bank WDI data base, 2009.

Figure 3.1



Income per Capita (PPP \$2000) in a Sample of Oil-Rich Countries a. Success Stories:

b. Oil Curse Stories



Figure 3.1 Continued c. New Entrants' Stories



Source: World Bank's WDI data base and Elbadawi and Kaltani (2008).

Figure 3.2 The Evolution of Income in Saudi Arabia, the OECD and Norway



Source: World Bank's WDI data base.

Therefore, though some oil-rich countries can in fact achieve higher levels of income and other economic and social indicators, they might, nevertheless, stagnate at levels that are far below their potential, given the resources they command. More importantly, many other less fortunate oil-rich countries may actually experience major post-boom growth collapses leading to absolute declines in their long-term income levels, often times associated with deep debt crises—the so called the "resource curse".

3.2 Resource Depletion and Negative "Genuine Savings"

"How can we reconcile the seemingly obvious fact that oil makes a country rich with the equally undeniable fact that few countries heavily dependent on the production of oil are as rich, in terms of per capita income, as many developed countries (Norway is an obvious exception), and, moreover, rarely even appear to be moving toward that goal?" (Geoffrey Heal, 2007: p. 156).

Unlike the former manifestations of the resource curse (e.g. economic concentration, extreme volatility, and post boom growth collapse), the full ramifications of the resource depletion nature of oil-based development are not yet fully appreciated. We refer to the above quote from Heal, who argues that the answer to the paradox he poses hinges on the fact that income generated from oil is quite different from income generated by other renewable sources. This has implications for the country's underlying wealth. Once oil-based development is recognized as a process of depletion of one form of asset (the stock of oil) to finance consumption and other forms of assets, the true wealth of Saudi Arabia, for example, may look very different from its apparent wealth as measured by the value of its oil reserves (Box 3.1).

Moreover, once we properly account for the depletion effect of oil in the national income accounts, most, if not all, natural resource exporters appear to be depleting natural capital faster than the rate at which they are building new types of renewable capital. That is, they are generating negative genuine savings. Therefore, even if some of these countries managed to achieve high income and welfare levels, they are not likely to sustain them in the future. This perhaps might explain why some relatively high income oil-rich countries, such as Saudi Arabia, have so far failed to break through to the income levels of developed economies (Figure 3.3). Worse still, for many other cases, the negative savings can be so extreme that an oil-rich country can experience an absolute post-boom decline in real income relative to the pre-boom levels.

To more concretely discuss the implications of resource depletion we highlight some salient aspects of the theoretical framework used to estimate changes in national income and welfare in non-renewable resource-based economies.²⁶

The model starts with the classical inter-temporal, optimizing problem of maximizing the present value of the benefits (welfare) V(S) that can be obtained from a given level of capital stock (the 'state') S_0 :

$$V(S_0) = Max \int_0^\infty u(C_t) e^{-\delta t} dt \qquad 3.1$$

This is subject to a set of constraints imposed by technology, institutions, and resource availability. The solution to this problem leads to the derivation of the following "generic" expressions for the change in welfare over time, which is exactly equal to "genuine savings" and the change in national income over time:

$$\frac{dV}{dt} = \sum_{i} \lambda_{i} \frac{dS_{i}}{dt} (genuine \ savings) =$$

$$\frac{d}{dt} (National \ Income) = \frac{d}{dt} \int_{0}^{\infty} \sum_{0} \lambda_{i} c_{i} e^{-\delta t} dt$$
3.2

Where λ_i is the shadow price²⁷ of capital good i, the stock of which is S_i and c_i is consumption of good i. The term for "genuine savings", records the total value of investments, net of resource depletion.

However, to more concretely bring out the implications of this framework for welfare and national income accounts in oil-rich countries, we further highlight the specific features of the solution for the case of an open economy with extraction capital.²⁸ In this economy, extraction is not costless and requires domestic investment I_d

Box 3.1: How "Rich" is Rich Saudi Arabia: the Paradox of Resource Depletion

Saudi Arabia, with proven oil reserves of 262.7 billion barrels, is for some a poster child for the oil producer as economic utopia. Yet a proper measurement of its true wealth suggests that it is not as rich as it appears. There are two ways of measuring the flow income of the average Saudi family:

- If Saudi Arabia could sell its oil (at \$60 a barrel) and invest the proceeds at 4%, then a typical family could earn \$100,000, which does not qualify as rich by Western standards.
- If, instead, Saudi Arabia can just extract as much oil as it can-about 8 million barrels a day—then the per family income is just over a quarter of the above, at \$28,000, barely above the US poverty line.

Moreover, the latter option is more realistic since Saudi Arabia cannot sell its entire oil reserves all at once without forcing the price down dramatically. Therefore, the wealth of Saudi Arabia is in fact more apparent than real. Three lessons that can be gleaned from this paradox:

- First, capital markets matter to oil-producing countries as much as the oil wealth itself, because their standard of living depends on access to these markets and how well they use them.
- Second, a resource-rich country, such as Saudi Arabia, is not necessarily "rich" in the conventional sense.
- Third, any measure of income or of wealth change for these economies must allow for the depletion of the natural resource stock.

Source: Heal (2007: p. 156-57).

in the extraction capital K_d but part of the net oil proceeds can be invested (at a rate I_f) in interestbearing foreign capital K_f with interest rate r. Assuming that the rate (R) at which the resource can

Figure 3.3

Genuine Savings in Mineral and Energy-Rich Countries



Source: World Bank (2006).

be extracted is bounded by the extraction capital $(R \le \alpha K_a)$ and that the output of the resource is proportional to the capital available for resource

extraction (that is, $\frac{dS}{dt} = \alpha K_d$), we have:

$$C = \alpha K_d - I_f - I_d + rK_f$$
 3.3

The solution of (3.1) subject to (3.3) and other technical conditions²⁹ results in the following expressions for net national income (NNP) and the change in welfare (the change in the state valuation function):

$$NNP = C + rK_f + I_f + I_d - \alpha K_d \qquad 3.4$$
$$\frac{dV}{dt} = genuine \ savings = \lambda_f I_f + \lambda_d I_d - \lambda_s \alpha K_d \qquad 3.5$$

The above two equations clearly imply that without accounting for the depletion effect of oil extraction we cannot properly measure the evolution of welfare and income in oil-rich economies. Hence, income and welfare comparisons like the ones contained in Table 3.1 might be grossly exaggerating the relative performance of oil-rich countries. Also the two equations highlight the critical importance of domestic investment and capital markets for converting the depleting oil stocks into renewable stocks of other types of wealth.

This framework has been extensively applied by the World Bank (2006) to compute the value of investment at shadow prices $(\sum_{\lambda} \frac{dS_i}{dt})$ for a wide range of countries.³⁰ The estimates suggest that most oil-exporting countries will have difficulty sustaining incomes and welfare in the longer run because they have generated negative genuine savings (Figure 3.3).³¹ Also using the same methodology, the World Bank (2006) estimates that as they deplete their natural capital--in this case mainly oil stocks--oil producers are substantially deficient in terms of accumulation of "replacement" tangible and produced capital (Table 3.4). However, major oil-rich Arab countries, such as Saudi Arabia, Libya and, especially Kuwait, appear to have accumulated much more substantial capital than the median for major oil producers. Nevertheless, they fall far short of the capital stocks of high income OECD and their total capital stocks are comparable to those of the upper middle-income group. Moreover, lack of data for most Arab countries precludes comparison with regard to types of capital, especially the intangible capital. As noted by Collier et al (2009), though it is subject to several conceptual qualifications,³² the robustness of the estimates generated by the World Bank's "green accounting" methodology on the investment deficit of resource-rich economies cannot be easily dismissed. In fact, based on evidence from twelve oil-rich countries, they find that while hydrocarbon revenues rose very sharply since 2000, there were no substantial increases in domestic investment shares in any of these countries. Moreover, they find no statistically significant association between hydrocarbon revenues and domestic investments in this sample.

So far we have discussed the manifestations and symptoms of the resource curse associated with post-boom, economic collapse. This is the lack of economic sustainability in the long term as a result of the failure to generate positive "genuine savings" as the oil stock is being depleted over time. Next we go beyond the question of how to the question of why through a selective but representative review of the literature.

Table 3.4

Capital Per Capita in and outside the Arab World: Produced, Tangible and Intangible

		Capital Per He	ad (2000 US\$)	
	Total	Intangible	Produced	Natural
Income Group (excl. oil countries)				
Low Income	7,532	4,434	1,174	1,925
Lower Middle Income	22,674	15,517	4,187	2,970
Upper Middle Income	76,538	51,001	16,831	8,706
High Income OECD	439,063	353,339	76,193	9,531
Major Oil Producers	22,952	2,359	7,937	
World (excl. oil countries)	95,860	74,998	16,850	
World	90,210	69,369	16,160	
Arab Countries				
Algeria	18,491	-3,418	8,709	13,200
Bahrain	112,254			
Comoros	8,030	5,792	1,270	967
Djibouti	16,248			
Egypt, Arab Rep.	21,879	14,734	3,897	3,249
Iraq				
Jordan	31,546	24,740	5,875	931
Kuwait	140,179			173,883
Lebanon	69,561			1,929
Libya	89,434			1,653
Mauritania	7,959	3,938	1,038	2,982
Morocco	22,965	17,926	3,435	1,604
Oman	49,445			68,835
Qatar				579
Saudi Arabia	60,605			
Somalia				
Sudan	5,681			2,951
Syrian Arab Rep.	10,419	-1,598	3,292	8,725
Tunisia	36,537	26,328	6,270	3,939
United Arab Emirates				
West Bank and Gaza	36,465			
Yemen, Rep.	5,911			

Table 3.4 **Continued**

		Capital Per He	ad (2000 US\$)	
	Total	Intangible	Produced	Natural
Best Performers				
Botswana	40,592	28,483	8,926	3,183
Chile	77,726	56,094	10,688	10,944
Indonesia	13,869	8,015	2,382	3,472
Malaysia	46,687	24,520	13,065	9,103
Norway	473,708	299,230	119,650	54,828
Worst Performers				
Angola				7,813
Nigeria	2,748	-1,959	667	4,040

Source: World Bank (2006)

Learning from the Resource Curse

Thinking through a viable strategy for avoiding the curse clearly requires a theory for explaining it in the first place. There are at least three approaches that attempt to explain the phenomenon--including those that emphasize the Dutch disease, governance deficit and rentseeking, and extreme volatility. However, as the mushrooming large N regression literature and several case studies have confirmed, it takes major policy failures in all of these dimensions to produce a post-boom collapse in the level of output, not just deceleration of growth.

4.1. The Dutch Disease Perspective

Well known in this literature, the name "Dutch Disease" describes the overvaluation syndrome of the Dutch guilder in the wake of a boom in the Netherlands due to the discovery and production of natural gas in the 1960s, which was blamed for the de-industrialization of the country. In general this hypothesis is based on the notion that the rise in oil revenues generates high overall demand, including for tradable imported goods and services a well as for a wide range of non-tradable home goods and services. The increased demand for the latter group must be met by increased domestic supply, because they cannot be imported. This, in turn, would require that resources and factors of production move from the non-oil tradable sectors (mainly agricultural and industrial exporting sectors) to the non-tradable home goods sectors (e.g., housing, urban consumer services, etc.)

How the Dutch Disease phenomenon works is

best explained through a graphical illustration of the classical three (oil traded, non-oil traded and non-tradable) goods model (e.g. Sachs, 2007).³³ Starting from the left panel of Figure 4.1, the economy's equilibrium E is at the point of tangency between the PPF (the production possibility frontier) and the consumer indifference curve II, and the economy is producing E_N amount of non-tradable goods (vertical axis) and E_{T} of total oil and non-oil traded goods. The line passing through the point of tangency is the real exchange rate (RER), which is generically defined as the price of non-tradable to tradable goods (p_N/p_T) . The steeper this curve the more depreciated (lower) the RER.³⁴ Now in the wake of an oil boom that is totally or mostly spent on consumption, the economy shifts to the new equilibrium depicted in the right panel of the figure. The PPF curve shifts to the right by the amount of the oil boom (H). Notice that there is no vertical shift in the PPF curve because no expenditure is allocated to productivity enhancing investments. The new equilibrium is now given by point E^{*}. The total (oil and non-oil) traded goods output is now increased from E_{T} to E_{T}^{*} +H. Moreover, as discussed above, output of the nontraded has also risen from E_N to E_N^* . However, very importantly, although the aggregate traded goods output has increased, the output of the nonoil traded goods sector has actually declined from E_{T} to E_{T}^{*} .

This process is mediated by the appreciation of the RER (reflected by a less steep curve through the new tangency point: E^{*}). The real exchange



Figure 4.1 The Geometry of the Dutch Disease

Source: Figure 7.1 of Sachs, 2007

rate (RER), as the relative price of non-tradable to tradables, goods, must appreciate to provide the necessary incentive for the inter-sector transfers of resources and factors to take place. As a consequence of the appreciating RER and, therefore, declining profitability for non-resource tradable sectors, agriculture and manufacturing shrink over time. This is essentially the basic story of the Dutch disease literature (e.g., Corden, 1982; Corden and Neary, 1984).

However, despite the appreciation of the RER and the subsequent squeeze on the non-resource tradable sectors, the pure Dutch disease effect cannot explain the overall collapse of the economy, as evidenced in the collapse of the GDP associated with the "oil curse" phenomenon. This is because, at the theoretical level, it is not clear why the declining share of the non-resource tradable sectors could not be compensated for by the rising share of the non-tradables. goods. However, the disease might become a true disease (i.e. an oil curse) under two very adverse conditions. First, if the squeezed activities assume some special qualities that could not be compensated for by the expanding resource and non-traded goods sectors. Therefore, an extended version of the Dutch disease thesis also assumes that exporting activities in general entail some dynamic properties, such as learning by exporting, and that manufacturing in particular generates increasing returns to scale (e.g., Matsuyama, 1992; Bigsten et al, 2002; Rodrik, 2007). Second, it also assumes that the RER appreciation is too extreme, relative to the RER

levels consistent with the long-term, sustainable path of the RER fundamentals. Again this is also a plausible assumption because the marginal impact of oil booms on the RER can be quite substantial, with elasticity estimates ranging between 40 to 50% (Korhonen and Juurikkala, 2007). As discussed in section 2, RER overvaluation can be very harmful to growth, especially for exports and economic diversification.

Empirical evidence suggests that, by and large, most oil-rich countries tend to experience large post-boom RER overvaluation, however it is important to make the point that an oil boom and the consequent RER appreciation (i.e. the Dutch disease) can be entirely consistent with equilibrium and need not, therefore, lead to an oil curse. Going back to the graphical model, let us assume that all or most of the proceeds from the boom were spent on infrastructure and other types of productivity-enhancing investments. In this case, the PPF experiences an outward (that is a simultaneous upward and rightward rather than just a rightward) shift. In this case the new economy's equilibrium is shown in the right-hand panel of Figure 4.2. Under this equilibrium (point E**) both the non-oil tradable as well as the non-tradable goods sectors will expand as a result of the increased productivity due to public investment. Of course the oil and the non-tradable sectors will also be positively influenced by the spending effect, which may or may not lead to RER appreciation depending on the net effect of the RER fundamentals and other short-run policy determinants that



Figure 4.2 The Effects of Investment via an Oil Boom

Source: Figure 7.2 of Sachs, 2007.

affect RER adjustment between equilibriums (e.g. Elbadawi and Soto, 2009). Moreover, to generate meaningful productivity growth, expenditure on infrastructure and other investments must be tailored to the absorptive and institutional capacity of the economy. This would necessarily require that investment expenditure should be phased in over time and may even have to be adjusted downward to absorb future negative shocks to oil revenues (e.g. Collier at al, 2009). Hence an effective domestic investment program is not likely to lead to incentive-incompatible and excessive RER overvaluation. This is because of the productivity growth effect, which permits an equilibrium RER appreciation. In addition, the expenditure profile associated with an effective strategy for building the domestic capital base of the economy must be consistent with long-term sustainability, given the extreme volatility of oil revenues.

Finally, Sachs (2007) argues that in oil-rich but extremely capital-poor economies, the productivity growth unleashed from large capital investment outlays in infrastructure could be so high that an oil boom might be associated with an RER *depreciation* not *appreciation* (Figure 4.3). Indeed, recent evidence for aid-recipient, low-income countries on the RER impact of surges in aid and remittances, which is similar to an oil boom in oilrich but low-income countries, suggests that the long-run effect of productivity is very substantial and that recent episodes of the aid boom have not been associated with large RER overvaluation (Elbadawi et al, 2008; Elbadawi et al, 2009). Also Mongardini and Rayner (2009: p. 15) in a recent IMF working paper find similar results for a sample of aid-recipient Sub-Saharan African countries and conjecture that their finding, "reflects the fact that transfers such as grants and remittances are generally channelled to productive investments that boost productivity or ease supply constraints in the non-tradable sector."

However, the extreme volatility of oil revenues appear to be posing the most serious challenge for these countries with regard to aligning domestic absorption to the long-term sustainable path that defines the RER equilibrium. In this context Hausmann and Rigobon (2003) develop a model to motivate their "inefficient specialization" hypothesis, which augments the basic Dutch Disease by accounting for the impact of the RER volatility on the non-resource tradable sector. They argue that for a resource curse to happen, the non-resource tradable sector must be substantially squeezed beyond a critical size or almost disappear and that the economy must be characterized by financial imperfections. They show that the near disappearance of the tradable sector will be associated with a highly volatile real exchange rate, because with very squeezed or no non-resource tradable sector, only expenditure switching forces will be at play in response to the shocks from the oil sector. In turn, financial imperfection would cause interest rates to rise with increased volatility of the real exchange rate. In their model, a vicious circle between greater volatility and interest rates, on the one hand, and lower investment in the tradable





sector, on the other, will lead to a steady squeeze of the latter until it eventually disappears and the economy specializes inefficiently in non-tradables. goods. They show that this economy will exhibit higher interest rates, lower capital and wages, and more depreciated exchange rates. Next we consider the volatility of the oil curse.

4.2 The Volatility Story

The recent literature on the development impact of external volatility finds that it has been negatively associated with a wide range of economic performance indicators, including growth, investment, income distribution, poverty, and educational attainment.³⁵ Given that oil prices (and hence revenues) tend to be very volatile, the tendency of the oil-dependent economies to experience frequent and oftentimes extreme volatility has been proposed as an alternative explanation for the curse. As noted by Hausmann and Rigobon (2003), one standard deviation shock to the price of oil (estimated at 30-35%) can generate an income shock as high as 6% of GDP in an economy where oil accounts for 20% of GDP. This is a very high volatility effect compared with the median shock for industrial countries (about 2%) or even developing countries (at 4%). The failure to cope with this extreme volatility, it is argued, has been the main factor behind the post-boom economic collapse experienced by most oil-dependent economies. Country experiences show how destructive the oil cycles can be when they are not properly managed. For example, Gelb and Grasmann (2008)

cite three such examples analyzed in the literature, including the Mexican debt crisis precipitated by the debt-financed spending spree against expectations of a longer oil boom that failed to materialize after 1981. Even more glaring examples were those of Nigeria (Budina and Wijnbergen, 2008) and Venezuela, which saw its per capita output reduced to half its real value following the end of the 1974-1980 oil boom (Hausmann, 2001).

In addition to its extreme volatility, oil prices (and hence revenues) are also highly uncertain because they don't follow a discernable trend and are, therefore, very difficult to predict (Figure 4.4). Recent studies analyzing the very long-run evolution of oil prices could not reject the random walk hypothesis. For example, Cuddington et al (2007) find that since the last century real oil and commodity prices have experienced one or more downwards structural breaks, but the data-generating processes seem to follow a random walk. The enormous uncertainty is reflected in the very large discrepencies between prices, especially looking to the distant future.³⁶ In a careful statistical study of oil prices, Hamilton (2008) forecasts oil prices for four years from 2008:Q1, when prices averaged \$115 pb. Based on his forecast he concludes that, "... we might have still "expected" the price of oil still to be at \$115 a barrel, though we would in fact not be all that surprised if it turned out to be as low as \$34 or as high as \$391!" (p. 4). Given these enormous spreads, discrepancies, Gelb and Grasmann calculate that, with oil valued at \$100 pb and futures prices ranging from \$50

Source: Figure 7.3 of Sachs, 2007.



Figure 4.4 Long Run Oil Prices: 1861-2008

Source: British Petroleum Statistical Review (reported in Figure 3 of Gelb and Grasmann)

to over \$150 pb, the differences between the two ends of the price spectrum would translate into 50% of the GDP for an oil exporter like Nigeria. These major discrepancies in oil futures appear to be among the reasons why only a few countries have tried to hedge oil price risks through this instrument.³⁷

In view of the uncertainty over future oil prices, Gelb and Grasmann develop a simple but intuitive model to analyze spending decisions when the post-boom outlook is very uncertain. The model is broadly guided by the permanent income hypothesis but also accounts for factors associated with absorption constraints and macroeconomic adjustment to negative and positive demand shocks. Their review of the literature suggests that there is considerable support for an asymmetric response, reflected by a kinked supply curve.³⁸ For example, Collier and Venables (2008) find that favorable changes in terms of trade do not have significant effects on growth, but negative shocks reduce output. In particular, for a typical African country, terms of trade loss of 10% of GDP reduces growth in the following three years by 3.6 percentage points. The implications of extreme uncertainty and asymmetric adjustments are very clear, conclude Gelb and Grasmann (p. 13-14): "under-spending when prices are high can incur a small loss of welfare, but overspending during a boom can be costly, especially if financed by borrowing." These implications are very prominently borne out in their model's simulations (Box 4.1).

An alternative view (Collier et al, 2009) argues that in oil-rich but capital-poor countries assigning high priority to rapid accumulation of the relatively high return domestic capital should dominate a purely permanent-income driven strategy. This strategy, therefore, is inherently less cautious than the one advocated by Gelb and Grasmann, at least until the domestic capital stock reaches the optimum target level. Nevertheless, the success of this strategy clearly requires that these countries should first attain the capacity to make effective investments. They should also be able to manage the political economy that would afford them the flexibility to undertake timely and adequate downward adjustments in new investments during bad time.times. However, this is not easy and depends on the quality of governance- both economic and political. We turn next to this issue.

4.3 The Governance Deficit and Rent-Seeking View

There is a vast literature on the role of institutions in economic development. However, the recent debates about the role of institutions in development, particularly in explaining long term income disparities across countries, has been mainly waged through the large N empirical literature. A summary of this literature suggests the primacy of institutions over other competing theories namely the trade and geography views. Four main conclusions can be gleaned from this literature (Elbadawi, 2005c):

- Institutions have direct effects on income, while geography doesn't. However, the latter influences institutions through settlers' mortality (Acemoglu, Johnson, and Robinson, 2001).
- Institutions determine long-term income levels, while neither policy, (including trade), nor geography, have a direct impact on income. However, geography determines institutions (Easterly and Levine, 2003).
- The most compelling evidence in support of the above conclusions is provided by Rodrik et al. (2002), who conclude that 'institutions rule'.
- However, Sachs (2003) shows that malaria transmission, which is strongly affected by

ecological conditions, directly affects the level of per capita income after controlling for the quality of institutions. This suggests that geography has not only an indirect effect on incomes, as suggested by the above literature, but also a direct effect as well.

In the case of oil-rich countries, the above conclusions could never be more relevant, as oil-rich but poorly governed countries are not likely to acquire the knowledge and the institutional capacity to take the right decisions regarding extraction, savings, and investments; to adopt the most appropriate macroeconomic framework for avoiding the Dutch Disease; or to better manage the rampant volatility that plague their economies. For example, the recent literature on managing oil economies has called for a strong role for the state in oil-rich countries to deal with oil corporations (e.g. Stiglitz, 2007). To maximize the returns for their oil resources, these countries need to deal effectively with the vastly more knowledgeable multinational oil corporations on relatively complex issues, such as the overall terms of agreements; proper evaluation of fiscal terms of contracts; or how best to auction oil rights.³⁹ However, as important as they may be, the most serious drag on these countries' ability to effectively manage the oil rents for development is not likely to be lack of knowledge or institutional capabilities. Instead, the most devastating aspect of the governance deficits in these countries is more probably the lack of accountability associated with the scale and nature of the rents generated from oil and other point-source natural resources.⁴⁰

Controlling for income and population size, recent empirical literature find that oil rents are significantly and robustly associated with lower public accountability in the oil-exporting countries of the Middle East. However, no significant differences were found between oil and non-oil countries on other aspects of governance (e.g. Ross, 2003; Sala-i-Martin and Artadi, 2002; World Bank, 2003). A comparison of the World Bank's six worldwide governance indicators suggests that while oil-rich countries, especially in the Middle East, tend to suffer from an overall governance deficit (Table 4.1), the latter is particularly glaring for the case of "voice and accountability" (Figure 4.5).

Admittedly, democracy is not a perfect correlate of good economic governance institutions; it has, nevertheless, been a very good indicator of



Figure 4.5 Voice and Accountability in the Arab world, Oil Countries and Non-oil Countries (2008)

Source: Worldwide Governance Indicators, World Bank, 2008.

Box 4.1: Optimum Saving and Spending during Oil Booms

Gelb and Grasmann ask a fundamental question: "How much should exporters be saving, considering the great uncertainty over whether high prices are temporary or longer-term?"

They note that, while the permanent income approach has been a useful guide to prudent spending decisions, it does not, however, provide guidance on how cautious spending patterns should be in the face of uncertain future revenues. In determining the right degree of caution, it is important to factor in absorption constraints and macroeconomic adjustment to positive and negative demand shocks. In this context there is considerable support for an asymmetric response, suggesting a supply curve that kinks in response to large booms and busts. For example, Collier and Venables (2008) find that favorable terms of trade shocks in developing countries do not have significant effects on growth, but adverse shocks reduce output. For a typical African country, a TOT loss of 10 percent of GDP reduces growth in the following year by 3.6 percentage points.

The implications of asymmetry, they argue, are clear: under-spending when prices are high can incur a small loss of welfare, but over-spending during a boom can be costly, especially if financed by borrowing.

Subscribing to the above, they develop a simple, stylized model to simulate optimum spending decisions during the oil boom, which among others, factors, has two key features:

- Simple, reduced-form model, where valuation is directly on "net" public spending, with utility represented by the log of net public spending to allow for diminishing marginal utility
- Net spending allows for two types of efficiency losses: first, beyond a moderate level of spending, the efficiency of spending begins to decline according to a quadratic function; and, second, asymmetric adjustment is assumed, with losses due to sharp declines in spending

Spending is optimized for two revenue profiles:

- A short boom, with oil revenues at 10, 30, and 10 percent of non-oil GDP in the pre-boom, boom, and post-boom periods respectively; and
- A long boom, with the revenue profile 10, 30, and 25 percent. Total spending is constrained to equal total income over the period

So, what does such a model suggest about spending decisions during a boom when the post-boom outlook is very uncertain?

- In the case of a short boom, the optimum is to spend 20 percent of incremental oil revenues (or just under 50 percent of total oil income) during the boom years and save the rest
- For a long boom, it is optimal to spend 80 percent of incremental oil revenues revenue, or about 85 percent of total income

Also, the losses due to misjudging the nature of the boom are asymmetric. Slow initial spending results in only a small loss of the potential value of the long boom; in the opposite case, over-optimistic initial spending results in the loss of most of the potential value of a short boom.

Moreover, each scenario is expected with a probability of 0.5. So the expected welfare-maximizing spending level, about 22 percent of incremental boom income, is not much higher than the optimal spending level for the short boom.

Governance in the Arab W	vorld and Com	iparators (2008						
Arab Grounings		Voice and Account- ability	Political Stability	Govern- ment Effec- tiveness	Regulatory Onality	Rule of Law	Control of Corruntion	Overall Governance Average
Oil-rich, labor importing countries (GCC + Libya)	max	-0.53	1.01	0.82	0.88	0.86	1.24	0.61
	median	-0.98	0.48	0.42	0.58	0.70	0.50	0.24
	Min	-1.90	-0.39	-0.84	-0.93	-0.65	-0.81	-0.78
Oil-rich, labor abundant	max	-1.05	-1.15	-0.50	-0.70	-0.70	-0.44	-0.77
	median	-1.22	-2.17	-1.20	-0.94	-1.22	-1.11	-1.35
	Min	-1.77	-2.69	-1.41	-1.36	-1.87	-1.49	-1.66
Non-oil middle income labor exporting	max	-0.40	0.29	0.35	0.34	0.49	0.41	0.08
	median	-0.94	-0.56	-0.37	-0.17	-0.11	-0.67	-0.53
	Min	-1.75	-1.94	-1.36	-1.17	-0.81	-1.13	-1.19
Non-oil low-income	тах	-0.43	-0.13	-0.97	-0.59	-0.54	-0.33	-0.64
	median	-0.92	-0.93	-0.98	-0.75	-1.01	-0.75	-0.87
	Min	-1.12	-1.01	-1.88	-1.51	-1.03	-0.80	-1.10

Notes: 4. Source: Worldwide Governance Indicators, World Bank, 2008. Max range = 2.5 and min= -2.5.

38

Table 4.1

how well an economy is managed. A careful analysis of the typology of political regimes in oil-rich economies suggests that: while unstable and factional democracies are as bad as unstable autocracies, well functioning, stable democracies are much better in managing oil resources than long-reigning, benevolent but nondemocratic regimes (Eifert et al, 2003). However, democracy has been a rare exception in the Arab world. Indeed, the region suffers from what has been famously coined by a widely quoted UNDP (2002) report as "freedom deficit". Despite the demise of the Soviet Union and major regional and world events, including the end of the 1970s oil boom, the Gulf wars, the worsening Palestinian crisis, as well as civil wars and other internal conflicts, no regime in the Arab world has extended the political franchise to the point where citizens could exercise effective control over public policy (Elbadawi, Makdisi, and Milante, 2010). Instead, democracy in other regions of the world has been steadily increasing, as demonstrated by trends in the Polity IV index (Figure 4.6). In the scale of the Polity score (-10: strongly autocratic to 10: strongly democratic), the East Asian and sub-Sahara African averages in 2003 were at -0.6 and 1.35, respectively, while the average for Arab states was -5.5, lower than the Arab average from 1960.

It is perhaps not surprising that, although overall the Arab world has remained stubbornly autocratic, the oil-rich countries have been extremely autocratic, with polity scores in the neighborhood of (-7, -10) for several decades.⁴¹ However, compared to unstable factional democracies in some oil-rich countries of Africa, their stability might be a virtue from the perspective of oil management, since sustainability of effective management is always a concern (Eifert et al, 2003).

Probing further, we organize our review around two questions that, in our view, have defined recent contributions to the literature: How might oil and other point-source natural resources undermine the quality of governance? And how effective can democracy and good institutions be in promoting development in the presence of oil rents?

Democracy and the quality of governance in oil-rich societies.

The theoretical literature on how oil and other point-source natural resources undermine the quality of economic and political governance is

Figure 4.6





Source: Polity IV dataset and Table 2 of Elbadawi, Makdisi and Milante (2010).

vast. For example, one strand of the literature that emphasizes the political economy of oil management argues that oil generates so much wealth that is also characterized by imperfect property rights. Therefore, agents in society find it much more profitable to engage in unproductive, rentseeking activities in order to appropriate that wealth rather than engage in activities to create new wealth. The imperfect property rights (or the common pool problem) stem from the fact that the benefits of oil wealth are internalized but the costs associated with overspending are diffused among many agents. Therefore, this leads to overspending on average because while a given constituency can internalize the full benefit of its overspending, it will only need to pay a fraction of the additional tax burden.⁴² In oil economies, where resource rents are high and taxes very low, this effect can be very powerful. Moreover, in a dynamic sense, the common-pool problem could also lead to the distorted allocation of spending over time. The mechanism for generating this is that each constituency would like to spend out the boom rather than smooth consumption over time, for fear that others will also spend. This is the so-called "voracity effect". 43

In many cases, the voracity effect also manifests itself in over-borrowing and debt overhang as countries use oil as collateral to borrow during the boom to finance accelerated expenditure of large infrastructure and other development projects. By and large, the project failure rate under these circumstances is very high, which leaves these economies exposed to very high indebtedness when the boom is over. The debt overhang associated with the oil boom has, therefore, been characterized as a plausible cause of the "oil curse" (e.g., Manzano and Rigobon, 2007).

Democracy, institutions, and economic performance in oil-rich countries.

Recent literature suggests that institutions are shaped by very long processes (e.g. Acemoglu, Johnson, and Robinson, 2001), so much so that they are likely to predate resource discovery (Smith, 2004). The premise is that institutions, especially political ones, are exogenous, however valid instruments do exist for them to be potentially endogenous. The literature has focused on analyzing the interactions between institutions and resource rents in empirical growth models. In particular, this literature assesses the effectiveness of democracy and other institutions of good economic governance in enhancing overall growth or stemming the consequences of the volatility associated with these rents.

Rodrik (1999) links the volatility story to the governance one and argues that the effect of external shocks on growth and economic performance in general is not just the outcome of the failure of adjustment policies in the technical sense; it also reflects the interaction of these shocks with latent social conflicts in society, on the one hand, and institutions for conflict management, on the other. In societies with deep social conflicts (for example, societies fractionalized or polarized along ethnic, religious, cultural, or economic class lines) and weak social and political institutions for mediating conflicts among social groups, Rodrik argues, that the economic costs of external shocks are magnified by the growth-retarding distributional conflicts that are triggered.44

Using a simple empirical proxy to test this hypothesis in a global cross-sectional data base, Rodrik finds that the interaction term

(shock x social conflict x lack of economic and institutional capacity)

is robustly associated with the collapse of growth

experienced by many oil-importing developing countries following the oil price hikes in the 1970s. More recently, Elbadawi (2005b) analyses the collapse of growth in the oil-dependent Arab world in 1985—94 (relative to the boom era of 1975–1984) and finds that it can also be explained by Rodrik's framework. Elbadawi's analysis suggests that the failure of the Arab region to sustain growth following the end of the oil boom in the mid 1980s is in part explained by its high dependence on a volatile resource base. But he notes that it is also due to the failure to develop the right social contract for generating high enough savings to permit insulating fiscal policy from the oil shocks affecting its economies.

Like Rodrik's analysis for the case of growth volatility and long-term sustainability, most of the received literature suggests that democracy and good institutions tend to promote a long-term level of growth and other aspects of development in oil-rich societies. For example, in a theoretical model of patronage politics in the context of resource wealth, Robinson et al (2006) show that good institutions may restrain this dysfunctional behaviour. Also Mehlum et al (2005) presents empirical evidence in support of the critical role of good institutions in resource-rich economies.

More recently, Collier and Hoeffler (2009) analyze an empirical growth model using global data from 1970-2001 and a new measure of resource rents.⁴⁵ They find that high natural resource rents and open democracy (as measured by Poliy IV) interact badly as determinants of growth, but checks and balances offset this adverse effect. Based on their results they conclude that democratization in resource-rich economies needs to emphasize strong checks and balances. Their empirical framework is motivated by a simple intuitive model that generates uncertain outcome for the growth impact of democracy in resource-rich economies, depending on whether or not democracy entails enough checks and balances to limit embezzlement of funds and hence maximize provisions for growth-enhancing public goods. Thus the key insight of their paper is modelling the determination of the checks and balances. They assume that politicians would like to heavily tax in order to generate patronage, but they will be deterred from doing so because high taxation provokes scrutiny. They assume that patronage expenditure, P, is determined by the product of the tax rate, t, and disposable income, Y, and the proportion of revenue which can be embezzled for patronage, e, which is simply given by a linear negative function of the tax rate: $e = \alpha(1-t)$. The decision problem for the corrupt politician is thus given by:

$$P^{\max} = \max_{wrt \ t} \alpha(1-t) \ t.Y$$

$$4.1$$

The following equations show the patronage maximizing level of the tax rate: $t^*=0.5$ and the maximum resources available for patronage (P*) and public goods (G*):

$$P^* = Y\alpha / 4 \tag{4.2}$$

 $G^* = (2 - \alpha)Y/4 \tag{4.3}$

Hence in this simple model, equations 4.2 and 4.3 describe the outcome of electoral competition with the restraint of endogenous scrutiny. Now when resource rents are introduced at a rate, r, available revenue becomes:

$$[t(1-r)+r]Y$$
 4.4

Note that the resource rents are not subject to scrutiny. The new optimization problem now becomes:

$$P^{r^{Max}} = \max_{wrt \ t} \alpha (1-t) . [t(1-r)+r] . Y$$
4.5

The patronage maximizing tax rate is given by:

$$t^{**} = (1-2r) / (2-2r)$$

This result suggests that the higher the revenue from resource rent, r, the lower the tax rate. However, though the resource discovery augments income, it is not clear whether or not it will increase the provision of public goods, depending on the scale of resource discovery, r, and the rate of embezzlement α . Collier and Hoeffler show this by analyzing the interesting case of a scale of resource discovery r=0.5, which will be consistent with t**=0. At this rate, both income and government revenue is doubled, because the state previously received half of national income (t*.Y=0.5Y) and now it receives all of national income. Also now with the tax rate driven to zero, and hence dampening the effect of scrutiny, the rate of embezzlement increases from $\alpha / 2$ to α . Therefore, the provision for public goods in the presence of resource rents is given by:

$$G^{**} = (1 - \alpha)2Y \tag{4.7}$$

Comparing G^{**} with G^{*} (in 4.3 above) suggests that the provision of public goods would worsen in the presence of resource rents if:

This is the upshot of these authors' model, which they use to explain some specific country experiences. For example, they argue that in a democracy with strong checks and balances α will be small, hence provision for public goods is likely to increase. This is the case in Norway, where strong checks and balances limit embezzlements and increase provisions for public goods even with small resource discovery, r. However, for Nigeria the likely high rate of embezzlement (as can be inferred from its ranking across a range of governance indicators) has probably worsened provision of public goods in the context of its moderate-size rate of resource discovery. Finally, Saudi Arabia would be the polar opposite to Norway, though with the same outcome, in a qualitative sense. Though the rate of embezzlements is likely to be very high in this country, again judging by the low governance ranking and the zero rates of taxation, the sheer size of the resource discovery is likely to improve provision of public goods.

Therefore, this simple model provides a nice empirical framework for testing not only the impact of contestable democratic politics on the development of resource-rich societies, but the specific features that might makes democracy effective in these countries.

4.4 Summing up

A review of the vast literature on the oil and nonagricultural natural resource curse suggests that there is no one encompassing theory that would explain the fundamental and truly exogenous causes of the phenomenon, the channels through which it operates, and the heterogeneity of country experiences. At the very least, it takes all three approaches combined to explain the oil curse. In this context, Hausmann and Rigobon's (2003) "inefficient specialization" thesis might be thought of as an attempt to synthesize the combined, if sequential, effects of the Dutch Disease, volatility, and specific institutional failures associated with the financial sector. In their model, the trigger comes from the TOR shocks and RER appreciation. These cause risk-averse investors to prematurely specialize in non-traded economic activities, which in the presence of weak financial institutions, further intensify volatility and economic concentration, leading to the eventual collapse of overall growth. This model, however, leaves out deep political economy channels that might be at work in explaining the phenomenon, such as the ex-post role of weak economic and political institutions in magnifying ex-ante TOR shocks in socially divided societies (Rodrik, 1999).

Therefore, despite the fact that major advances have been made by both the theoretical and empirical strands of the literature, several major issues remain open for further research. First, though the majority view is that there is robust, if conditional, evidence on the existence of the curse, there are still some who continue to question its existence (Lederman and Maloney, 2008). Second, there is now fairly robust evidence that the curse is likely to be conditional on severe governance and human capital deficits, which also appears to explain the heterogeneity of country experiences. Third, the literature is still unfolding regarding which channels produce the curse phenomenon in countries suffering from the twin deficits. Fourth, though most of the theoretical and empirical literature seems to suggest that factors influencing institutional quality and human capital are likely to be causally prior to oil discovery, reverse causation cannot be ruled out.

These issues will be revisited in more detail in the last section, which discusses future research. The received literature has, nevertheless, provided very important insight for discussing strategies for escaping the oil curse. This will be the subject of the next section.

Escaping the Oil Curse: The Macroeconomic Framework

n the above literature review, we briefly discussed the strategic decisions regarding sav-Lings and investment of oil rents. However, regardless of the chosen strategy, a complementary macroeconomic framework for medium-term management is required. The framework should be guided by two strategic objectives: optimum savings, consistent with the sustainability considerations of the oil and mineral economies, which would require much higher average saving ratios (over the cycle), compared to the non-mineral economies. And, it should also be guided by stable, optimum expenditures, consistent with a long-term development vision, such as the poverty reduction strategy (PRS) and other Millennium Development Goals (MDGs). For some oil and other mineral-exporting countries the compelling reality of the eventual extinction of the oil and mineral resource in two or three generations (e.g. Norway, Oman, and Gabon) requires that substantial portions of the oil and mineral rents should be saved and prudently invested to prevent a sudden collapse of social welfare when the oil and mineral rents cease to exist. For many other countries, however, this may be a relatively abstract eventuality, with oil and mineral reserves extending for fifty or more years. Nevertheless, even for the latter countries it makes good economic sense to diversify sources of national income by saving for future generations and investing in economic diversity. Moreover, aside from the long-term objectives of saving for future generations, a key objective of short-to-medium term macroeconomic

management should be to insulate fiscal and monetary policy from the volatility associated with the oil and mineral cycles.

However, the evidence suggests that most emerging market economies, especially those dependent on oil and minerals, have not been able to develop effective counter-cyclical stabilization policies. In addition to the political economy and institutional weakness, this evidence has been also linked to the choice of inappropriate monetary and exchange rate regimes (e.g Sester, 2007); domestic and financial imperfections (Caballero, 2002; Caballero and Krishnamurty, 2000); and recurrent credit constraints in world markets and "sudden stops" (Reinhart and Calvo, 2000). The failure to undertake effective counter-cyclical fiscal and monetary policy results in low credibility and dynamic policy inconsistency (Calderon and Schmidt-Hebbel, 2003).

However, there are also some notable successful experiences that might be a useful guide to the oil-rich Arab world. There are three issues that arise in this context. First, there is now an emerging consensus that because they complicate adjustment to external shocks, the costs of fixed exchange rate regimes, especially hard pegs, might outweigh their benefits for oil-exporting economies. Second, that rule-based fiscal policy, supported by a sufficiently flexible exchange rate regime, can be an effective counter-cyclical fiscal institution. Third, political economy considerations are very critical for the success of special fiscal institutions, especially long-term sovereign wealth funds (SWFs). Discussion of these two issues follows.

Fiscal Policy Rules.

The centerpiece of the macroeconomic framework is fiscal policy. However, effective counter-cyclical fiscal policy seems to be difficult to come by without credible and predictable fiscal institutions, guided by coherent objectives and underpinned by transparent processes. One such institution is the rule-based fiscal policies, recently adopted by some countries to eliminate the pro-cyclical bias of fiscal policy and to allow automatic stabilizers to work freely along the business cycle and keep surpluses gained in good times out of reach of the political establishment. The "structural fiscal balance" is the key concept driving these rules. In the case of the Chile, for example, the concept is centered on the following structural revenue formula (Garcia et al, 2005):

$$SB_{t} = OB_{t} + \left(T_{t} \cdot \left\{\frac{Y_{t}^{trend}}{Y_{t}}\right\} - T_{t}\right) + Coppersales(P_{t}^{trend} - P_{t}^{FOB})$$

$$5.1$$

Where in any given year, t, SB stands for structural revenue adjusted from observed revenue, OB, through two channels. First, the tax revenue plus pension contributions, T, are adjusted for the output gap; and second, the copper sales are adjusted for the difference between trend and actual FOB prices of copper. The long-term path of nominal GDP (Y^{trend}) and the sustainable trend prices of copper (Ptrend) are estimated by an independent panel of experts appointed by the government. While obviously the technical challenges of estimating these pivotal indicators should not be underestimated, the real challenge for most oil and mineral exporting countries would be the institutional design associated with the process. Next, public expenditure, PE, is given by:

$$PE_{t} = SB_{t} - (0.01)Y_{t}$$
 5.2

Thus in the Chilean structural fiscal policy rule, only revenue (not expenditure) is adjusted for the business cycle in the copper sector as well as for the overall output gap relative to its long-term trend. Also the rule imposes a "structural fiscal surplus" (SB-PE) equal to 1% of GDP. Note that when OB>SB, funds will flow into the country's sovereign fund, while if OB falls short of SB, the deficit will be financed by contracting new debt or withdrawing from the savings fund.

The Chilean fiscal surplus rule has been credited for flexibility, counter-cyclicality, and growthorientation. Since the Chilean structural balance reflects the level that would be observed if the GDP were at its estimated trend level and the international price of copper were at its estimated average long-term level, it should, therefore, eliminate the cyclical effects and temporary shocks of both the GDP and the copper price. As explained by Schmidt-Hebbel (2007), although government revenue follows the business cycle, the government must set its expenditure in a way that tracks the path of structural revenue. And, the constant structural fiscal balance implies that during booms, actual surpluses are higher (and will be saved) and during recessions, actual surpluses are lower, which would require drawing from the copper savings account. Hence, the Chilean fiscal rule implies an active counter-cyclical fiscal policy, which was very clearly borne out by the evidence.

The benefit of the presumed effectiveness of the rule as an instrument for counter-cyclical fiscal policy is manifest in the stability of public spending in investment and key social sectors, especially in education and innovation. In turn, this has led to relatively stable output and inflation, despite the high volatility of the copper sector. Moreover, the ensuing fiscal policy predictability has strengthened government solvency and raised its creditworthiness and improved the country's sovereign risk rating. Also very critically, the fiscal rule is also linked to Chile's ability to limit the Dutch disease consequences of the recent (2005–08) massive copper boom (Garcia et al, 2005; Schmidt-Hebbel, 2007).

However, it is not clear if the Chilean fiscal policy rule (and the stabilization program associated with it: the 1% fiscal surplus) can be successfully adopted in other countries, where institutional capacity or democracy standards are not as good, such as the Arab oil-rich countries. The Chilean fiscal rule was a culmination of a long history of fiscal prudence reflected in the achievement of surpluses on a cash basis for twelve consecutive years and the creation in 1987 of a fiscal revenue stabilization fund financed by proceeds from the copper sales. Therefore, while it may be argued that fiscal institutions can be effective instruments for overall institutional development, they may need a minimum level of good governance to be successful on a sustainable basis. Indeed, available econometric literature suggests that, controlling for institutional quality, there is no robust evidence that stabilization and savings funds lead to better management of natural resources (Davis et al 2003; Fasano, 2000).⁴⁶

Moreover, another debatable issue is the adoption of the fiscal surplus target rate of 1% of GDP. In the first place, why a structural surplus of 1% and not a balanced budget? The case for the former has been argued for the case of Chile on the grounds that global capital markets tend to have a lower tolerance for public debt in emerging market economies, especially those susceptible to large commodity shocks. Moreover, net asset accumulation over time by the government might be necessary to meet public sector commitments and contingent liabilities that grow at higher rates than the fiscal revenues (Garcia et al, 2005).⁴⁷ However, in 2008 the Chilean government reduced the fiscal surplus target rate to 0.5%, which has been assessed as desirable from a business cycle perspective, because the earlier rate would have required significant asset accumulation that could have been accomplished at the expense of greater volatility in fiscal instruments and hence GDP (Kumhof and Laxton, 2009).

Finally, another issue of interest is that while the structural fiscal surplus approach might be construed as premised on PIH, the principle might be consistent with the accelerated domestic capital accumulation view of Collier et al. For example, the choice of the size of the target surplus could be linked to capital investment requirements in the domestic economy along the lines of the Collier et al framework. However, more fundamentally, trend output could be estimated based on the concept of "optimum" or "desired" capital stock. For example, trend output is estimated in the Chilean fiscal policy rule by a Cobb-Douglas production function with constant returns to scale:

$$Y_t = A_t K_t^{\alpha} L_t^{1-\alpha}$$
5.3

And, the capita stock is constructed by the commonly used formula:

$$K_t = K_{t-1}(1-d) + GFCF_t$$
 5.4

Where d is the annual depreciation rate and GFCF is the gross fixed capital formation corrected for intensity use. When the oil-rich country is substantially capita-poor, relative to a notional optimum level that is consistent with economic diversification and long-term sustainability, a high enough GFCF can be used. This will lead to a higher estimated trend GDP and structural balance, given the desired level of public stock.

The Political Economy of Natural Resource Funds (NRFs).

Several arguments can be made in favor of NRFs as instruments for augmenting transparency, accountability, and good governance, especially under conditions of weak governance, limited transparency, and when the executive branch is not accountable or subject to effective parliamentary oversight (Gelb and Grasmann, 2008). Nevertheless, there appears to be a consensus that resource funds are neither necessary nor sufficient for sustaining good macroeconomic management in volatile oil and other mineral economies (Davis et al, 2003). Yet there has been a mushrooming of NRFs, and 23 have been recently identified, including 14 owned by oil exporters (Rietveld and Pringle, 2007). Humphreys and Sandbu (2007), who argue that the economic case for resource funds is "surprisingly" weak, suggest that the rationale for NRFs is political.

Indeed, the agenda for oil management entails fundamental decisions that have major consequences for inter-generational entitlements to the rents as well as the temporal distributional issues among various groups in society. Undoubtedly, this is a deeply political strategy and success would, therefore, require a political process, a social contract, for organizing consensus and buy-in on the part of a wide network of stakeholders in the society in question. A credible social contract must be a product of a legitimate participatory political process and must be fiscally viable. The latter requirement is in turn dependent on the quality of the economic governance: effectiveness, accountability, and transparency of economic policy. This is generated by the political system and the technical soundness of the economic strategy for managing the oil rents. The most difficult issue is what determines the political process underlining the desired social contract. The prevailing political institutions prior to the oil boom (or before a country become an oil producer) are important determinants of how oil rents are likely to be managed. For example, the fact that Norway was already a well functioning democracy has a lot to do with the stellar success of this country in managing its oil rents. So politics affect how oil is managed.

On the other hand, oil also affects politics. The presence of large and concentrated rents in the hands of the public sector can weaken agencies of restraint in the political process and transform the social and political institutions of an oil country into what is referred to in the political science literature as the "rentier state" (Karl, 1999). Moreover, and even worse, another strand of the literature finds that natural resource rents are associated with increased risks of civil wars (e.g., Collier and Hoeffler, 2004). The corrosive effect of oil rents is most dire when the political process is still unfolding or dominated by autocratic regimes, where it could give rise to a self-reinforcing "legacy of overly-centralized political power, strong networks of complicity between public and private sector actors, [and] highly uneven mineral-based development subsidized by oil" (Karl, 1999 page 34).48

The economic and political keys to successful management of oil rents are: political stability; a measure of political legitimacy; long policy horizon; high savings; strong competitiveness underpinned by a powerful non-oil constituency. Using these performance criteria as a framework to assess the experiences of a variety of political regimes in oil exporting countries, Eifert et al. (2003) construct an exhaustive typology of political regimes and associated performance (see their Table 2). A summary of the main findings follows:

• Successful regimes: There are very few success stories outside mature democracies such as Norway or other old democracies in resource-intensive countries like Australia and new ones like Chile. However, partial (non-factional) democracy is very promising, with Botswana and Indonesia as notable examples. Moreover, reformist "autocracy" also has some successes (UAE, Chile under the military). We will argue, however, that the likelihood of stable, reformist development autocracy in socially fractionalized Africa or

the Arab world is very remote. This is because autocracies in such societies are likely to be captured by sub-national entities along ethnic, cultural, or religious lines. The experiences of the historically benevolent authoritarian Asian bureaucracies (e.g. the Republic of Korea, the Province of Taiwan and, more recently China) are not necessarily transferable to the Arab world.

• *The failures:* Paternalistic autocracy has some measure of legitimacy, a long planning horizon, and stability. Its undoing is lack of transparency, low competitiveness, and unsustainable fiscal policy. This leaves two more political regimes, namely predatory autocracy and fractional democracy, which are simply non-starters because of their limited political legitimacy, short-term policy horizon, and little or no savings.

However, the most relevant research and policy issues, especially for the oil-rich Arab world, are not likely to be illuminated by stressing, as we did above, the value of good governance and democracy for successful management of oil rents. Instead, as Humphreys and Sandbu⁴⁹ recommend, research should start by trying to understand the political incentives that make politicians unwilling to abide by the economically optimal policy rule. In this context they argue that the design of NRFs should not only approximate optimal fiscal policy but also create political incentives (or at least attempt to neutralize the political disincentives) for respecting the policy. They ask the question as to whether NRFs can be used to realign incentives in the political process in a way that reinforces sustainable commitment to the optimum policy rule. They develop a theoretical model of interest-group politics with power rivalry to systematically analyze this issue. They identify three broad sets of responses:

- First, NRFs can be designed to make discretionary finance more difficult in the future, and this will also reduce the incentive for reigning politicians to overspend now.
- Second, broadening the decision-making authority will lead to greater predictability and moderation in future spending and will also have a similar effect at present.
- Third, NRFs can have beneficial effects by en-

hancing transparency through educating voters about government successes and abuses in managing resource wealth.

The analysis in the previous sections makes clear that the development of the Arab world crucially depends on better management of the oil sector. However, there is very little research on the policy challenges and development consequences of oil for the region. It is not surprising, therefore, that while the received literature on the economics and politics of oil management is still unfolding, the knowledge gap is particularly severe in the Arab region. In particular, a critical mass of rigorous policy research for better understanding and, hopefully, better informing development policy in the oil-rich Arab world is lacking. In this context we propose three research projects that should be accorded high priority because, in our view, they constitute cutting-edge research in the literature that is also relevant to the development policy agenda of the region.

The two proposed research projects are: macroinstitutional strategies for escaping the oil curse, and economic policy and export diversification with special reference to the Arab world.

Issues for Future Research: Escaping the Oil Curse

This project would address two broad sets of issues: understanding the oil curse and building on that understanding to research strategies for escaping it; in other words, strategies to harness the oil resource for the long-term development of the Arab world.

Understanding the oil curse. There is fairly robust evidence that the curse is real though a minority opinion still questions its existence when resource abundance (such as net resource exports per capita) measures are used instead of the resource dependency measures (such as resource exports/ GDP), which has been the staple of the "large N" econometric literature on the development impact of oil.⁵⁰ However, other contributions find that even when measures of resource abundance are used, the curse seems to hold against a variety of robustness checks.⁵¹ Therefore, there appears to be a near consensus that the curse is, indeed, real. However, almost all the empirical literature finds that the existence of the curse is conditional on bad governance. This finding also resonates well with the theoretical strand of the literature. More recently, in an ambitious but carefully done paper, Collier and Goderis (2008) use a panel cointegration empirical growth model to analyse two issues that, in our view, constitute the point of departure for further research on the oil curse question.

Firstly, the report finds robust evidence that the change in the non-agricultural export price index is positively associated with growth, but the level of the index has a strong negative growth effect. This suggests that commodity booms have positive short-term effects on output but adverse long-term effects. Therefore, the curse operates the in the long-run.

- The distinction between the short and longterm effects is an important area for future research
- Specifically, the analysis of the short and longrun effects should be based on an even more flexible model that embeds panel co-integration, such as the Pooled Mean Group (PMG) estimator (see appendix II):⁵²
 - Like panel co-integration, the PMG imposes the same long-run coefficient across countries
 - However, it is more flexible than the former because it allows the short-run dynamic effects to vary across countries
- Analyzing the distribution of the short-run effects of the oil boom might provide useful insight on how the boom interacts with country characteristics even in the short-run

Secondly, these authors also find that, conditional on bad governance, controlling for the real exchange rate, public consumption and private consumption as channels of the resource curse, higher commodity prices no longer have a negative, long-run effect. The empirical significance of these channels corroborates recent political economy literature, This literature predicts that

permanent resource booms when government accountability is lacking allow politicians to expand public sector employment or to directly boost private consumption to enhance their popularity (e.g. Robinson and Torvik, 2005; Robinson et al, 2006). In addition to these distributional aspects in resource economies, another strand of the literature suggests that bad governance also discourages overall savings and overall spending, which is reflected in appreciated real exchange rates (e.g. Matsen and Torvik, 2005). Moreover, the significance of the real exchange rate channels also coheres with the recent literature on its role as an instrument in the development strategy for economic diversification, sophistication and growth (see section 2).

However, the above empirical evidence, while intriguing is, nevertheless, not yet sufficiently corroborated by other work, which suggests that this should be on the agenda for future research. Also, the evidence on the real exchange rate raises some further conceptual and empirical issues that should be considered as well:

- Assess the empirical relevance of these and other channels against a variety of robustness checks
- Distinguish between real exchange rate appreciation that may or may not be consistent with RER overvaluation:

- In principle, only the latter should be the appropriate channel for the kind of extreme Dutch Disease associated with the oil curse.

Thirdly, this literature emphasizes a particular aspect of bad governance: lack of checks and balances (e.g. Humphreys and Sandbu, 2007; Collier and Hoeffler, 2009). However, given the glaring democracy deficit in the Arab world, the research should address the following questions:

- What is the likelihood of attaining the required standard of checks and balances for averting the oil curse under conditions of extreme autocracy; or,
- Is it possible to design an economic governance system that is incentive-compatible with the required level of checks and balances in an otherwise essentially non-democratic but fairly stable political regime

The macroeconomic framework. The mediumterm, macroeconomic framework for oil management is inextricably linked to the underlining savings-investments strategy for oil rents. For example, most rule-based, fiscal institutions (such as the Chilean structural fiscal surplus rule) and natural resource funds (most notably the Norwegian and the GCC sovereign wealth funds) are based on variants of the permanent income hypothesis. More recently, however, Collier et al (2009) argue that the PIH-based strategy is not appropriate for oil-rich but capital-poor economies. Instead, they propose an alternative savings-investments strategy that would target an accelerated build-up of domestic capital stock. Under this strategy, the larger portion of the oil savings generated during the boom would be used to finance domestic investment rather than being invested in the lower returns global capital market.

This research could contribute to this literature by further analyzing the following issues:

- Compared to the PIH, does the proposed domestic capital strategy offer a more balanced social welfare approach between current and future generations?
- Also, might it achieve a better balance between short-term stabilization and longer-term structural transformation?
- And, what are the political economy constraints that might impede scaling down investment during oil busts?
- Moreover, might this strategy be more appropriate for the oil-rich, labor abundant Arab countries, given that their economies are likely to be deficient in terms of domestic replacement capital?

However, the research on the macroeconomic management of oil in the Arab world remains relatively undeveloped, even without accounting for the complex dynamics entailed by the alternative Collier et all propose. It is pertinent to mention that the current crisis has shown that fiscal policy is more important than previously thought. However, while developed countries have been effective in deploying counter-cyclical policy in the past, their developing counterparts have not been that successful. Especially in commodity exporting countries, fiscal policy has been pro- not counter-cyclical

- Pro-cyclical, discretionary fiscal policy dominates in resource-rich countries
- Instead, rule-based fiscal policy has been relatively effective as a counter-cyclical policy instrument in developing countries
- However, only a handful of democratic and well managed, natural resource-rich countries adopt rule-based institutions (e.g. Botswana, Chile, and Norway)

Based on the above lessons from the received literature, this research should address the following important political economy questions for the Arab world:

- Given the democracy deficit and lack of accountability and transparency that characterize economic governance in the Arab world, are rule-based institutions feasible in this region, or;
- Instead, might these rule-based institutions be the vehicles for addressing, or at least avoiding, the above wider governance problems that afflict the Arab world
- In particular, how useful might the proposals for encouraging transparency and respect of established expenditure rules on the part of policy makers be for the design of these fiscal institutions, especially NRFs.⁵³

Notwithstanding these political economy challenges, fiscal policy models of medium-term expenditure smoothing still need to account for productivity growth, different rates of return on government debt, and on financial, physical, and social investment (Leigh and Olters, 2006). These issues will be even more important under the proposed domestic investment strategy. Other issues for future research include addressing the inherent arbitrariness of fiscal rules (e.g. whether to target deficit, spending, or debt); how high the benchmark should be (e.g. why should the Chilean fiscal surplus be fixed at 1% rather than, say, 0.5% of GDP). Moreover, under the domestic investment strategy, some desired level of capital stock will be the main target for fiscal expenditure. Then there is the issue of the tradeoffs between simplicity and flexibility in the design of fiscal rules. Finally, there are also more practical policy issues for the oil-rich, labor-importing economies of the GCC that would require further research:

- Given the limited capacity of these economies to absorb major new investment in infrastructure, it would be interesting to assess the social returns for these countries of investments in infrastructure and other sectors in the regional Arab economies (or African and Asian economies for that matter); and,
- Under what conditions are such investments likely to be a better alternative strategy for generating higher returns but also for diversifying risk, given that several SWFs from these countries have been hit hard by the current global financial crisis?

The other component of the macroeconomic framework is monetary and exchange rate policy. There is a large literature on the optimum monetary and exchange rate regimes for developing countries, including oil-rich countries. However, there is little research on this literature concerning the Arab world, despite the fact that the choice of a monetary and exchange rate regime is considered one of the most important macroeconomic policy decisions a country must make. This can be articulated for the case of the GCC countries along the following lines:

- Except for Kuwait, which switched to a basket peg in 2008, all other oil-rich GCC countries maintained their currency peg to the US dollar
- With the perfect capital mobility that prevails in these countries, the pegged regime means that they do not have an independent monetary policy
- However, the rationale for this choice is that by giving up exchange rate flexibility and monetary policy independence these countries can import US monetary stability under free capital mobility

However, recent experiences have shown that the pegged exchange regime might have come at a huge cost to these countries. This is because the often divergent fundamentals between the US and these economies complicate their capacity to undertake counter-cyclical macroeconomic policy:

 When the US economy slows down the Fed will likely adopt expansionary monetary policy through cutting down interest rates, such as during the current global crisis

- Should this coincide with an oil boom the appropriate response in the GCC economies should be one of monetary tightening
- However, to maintain the currency peg, their interest rates cannot significantly diverge from that of the US

- This was the situation in 2008, which led to the inflationary spell in the GCC

- Instead, there had been previous episodes when monetary restraint in the US precipitated deflationary pressures on the GCC economies

This research, therefore, should analyze alternative exchange rate regimes and their technical and political economy feasibility for the GCC. As mentioned above, there is very little research on this area regarding the GCC countries,⁵⁴ but a few ideas have been floated in the literature for oil and commodity exporters, including:

- A basket peg that includes the price of oil together with the dollar and other major currencies (Setser, 2007); or,
- Simply pegging to the price of oil (or the main commodity export) as in Frankel and Saiki (2002) proposal

It has been noted that the GCC countries have so far been able to conduct counter-cyclical macroeconomic policy during the current downturn because they have a tremendous fiscal space provided by their accumulated reserves and large SWFs. However, this begs the question as to whether other regimes might be better for these countries from the viewpoint of counter-cyclical macro policy.

The polar opposite to the fixed regimes is pure floating, which remains largely confined to developed and a few developing countries with exceptionally strong macroeconomic management capacity, such as Chile. Even in the Arab countries that have recently adopted inflation targeting regimes, such as Egypt and Tunisia, these regimes remain substantially managed. In fact, in a paper discussing exchange rate regimes for the MENA region, John Williamson proposes intermediate regimes for this region.⁵⁵ The point of departure, he argues, was that an appropriate exchange rate and monetary regime should not be judged purely on whether or not it's prone to crises. Based on this criteria alone, floating regimes will have no contenders. However, he argues that floating will lead to significant long-term misalignment and will, therefore, have adverse consequences for growth and development. He concludes that for developing countries this should not be an acceptable bargain. Instead, an appropriate regime should be one that is capable of minimizing the danger of provoking speculative attacks while helping to avoid real exchange rate misalignment.

This project, therefore, should also evaluate existing exchange rate regimes in GCC and non-GCC Arab countries, including analyzing:

- The typology of exchange rate regimes in the Arab region and their relative performance in terms of counter-cyclical policy and real exchange rate competitiveness
- The underlining political economy behind the choice of the exchange rate regime—for example, the presence of influential exporters might lead to the choice of a managed float with a bent toward RER depreciation; or a "flexible" rather than "strict" inflation targeting that also account for the RER as a secondary target (an extended Taylor Rule).

Issues for Future Research: Economic Diversification

e start with some thoughts on the question of whether and how to diversify oil economies, with special reference to the economies of the Middle East, and then suggest some possible issues for research. Globally, diversification has been an objective of many primary exporters for many years. As a whole, developing countries have been spectacularly successful; whereas 80% of their exports were primary commodities in the 1960s, today 80% are industrial products. With only a few exceptions, those countries identified as particularly resource-rich, whether hard-mineral or hydrocarbon-based, have been less successful, and this extends to countries in the Middle East. Whether this reflects the difficulty of countering the strong pull towards primary specialization impelled by current comparative advantage ("sectoral Dutch Disease") or political and governance weaknesses due partly to rent-seeking ("institutional Dutch Disease") or other factors is an open question.

Before considering specific macro and micro policies and research questions, a few basic questions should be considered:

Why Diversify? It is not automatic that countries have to move away from resource-based sectors to develop. Some now-developed countries such as Australia base their economies on natural resources, and Finland, the US, and many other countries did so for an extended period. Some studies, such as Lederman and Maloney (2008), suggest that with good-quality complementary inputs of human and governance capital, resources provide a positive base for development.

In advocating policies to diversify, we should be very clear on the objective of these policies. Is it for future income growth: to create domestic sector capacities with dynamic learning that can substitute for oil when reserves decline, or for insurance against technology change (such as clean nuclear reactors and improved batteries for vehicles) which would offer the world substitutes for oil? Is it for asset diversification -- perhaps it is not politically or strategically acceptable to accept a future where almost 100% of the country's assets will shift from below ground to banks abroad. Is it to create jobs? If so for whom, at productivity and skill levels corresponding to what level of expected future income? And why not jobs in the non-traded sectors? These differences are important; for example, if insurance ranks high in the set of objectives, a country might expect to pay a long-term cost for following this option even if specialization in oil is expected to be the most productive activity for a long time to come.

Diversification policy is often equated to industrialization. This raises the question of whether there is anything special about manufacturers, or can other sectors such as services play a similar role?

These questions are important for Middle Eastern resource exporters, especially because of the large differences between the two types of countries:

- Lower-income countries with large populations relative to resources, and
- Higher-income countries with very small populations, in particular the GCC countries. Saudi Arabia is usually included in this group but fits only to some degree, since its population is quite large and growing rapidly.

The first set of countries face more "normal" problems. They are relatively low income, their oil reserves are limited, their populations are young, growing, and seeking employment opportunities and avenues for on-the-job learning. Even if they are currently oil exporters, they need to think about alternative areas of comparative advantage in a relatively short-term context. In some cases a combination of population and income growth could erode their oil exporter status, as has happened in other countries such as Indonesia.

The issues will be very different for the latter GCC-type countries. They have many years of low-cost reserves. Some 80% of their labor force is foreign, and most nationals work in public employment. A form of industrial policy which involves combining domestic investments financed by oil income with imported industrial labor is essentially another form of rentier policy, an alternative to investing in such sectors in a foreign country. As high-income countries, they might also want to target high-productivity service sectors or sophisticated components of production chains as nodes of diversification, so raising the emphasis on investing oil income in the creation of high-level skills.

One interesting research question is therefore to attempt to go back to basics to understand the particular benefits that diversification policy is trying to obtain for the different types of countries.

This question is closely related to the nature of the externality or market failure which diversification policy is aiming to identify and address. Noland and Pack (2008) motivate industrial policy in several ways: infant industries, coordination failures across sectors, first mover costs, and rentshifting in oligopolistic sectors. As discussed in the extensive survey by Harrison and Rodriguez-Clare (2010), it is also necessary to consider the criteria needed for industrial policy to be "a good investment" over the long run. Even if successful in the sense of creating a more diversified economy, diversification policy can be a losing proposition in strictly economic terms. Some apparently successful interventions, such as South African support to its automotive sector, have been criticized on such grounds (Flatters, 2005), and Noland and Pack note that industrial policies may have been more successful in shifting resources around and changing sector composition than in increasing productivity and incomes.

Diversification, Oil or Other Sectors? A second general question is diversification within hydrocarbons versus diversification to sectors outside and unconnected with this sector. Many resource-rich countries' first move away from primary production is towards resource processing, as well as perhaps developing capabilities in supplier industries. Some countries in the Middle East, in particular Saudi Arabia, have made massive investments in petrochemicals, including developing substantial private sectors. These tend to be highly capital and skills intensive, and the record of heavy industrial investment in resource exporters is a mixed one (Auty, 1990). But they do provide increased opportunities for domestic supplier industries, as well as an incentive to develop the technical skills of nationals in the related areas. An interesting research question could be to compare the benefits from the hydrocarbon-based industry relative to those sought from diversification in the wider sense.

Country-Based or Regional Approaches? While most of the literature on industrial policy focuses on the choices for individual countries, collective regional action can be important in some contexts, for example in Africa with many small, landlocked countries. Another question for research could be to ask what specific regional policies-including those related to the sharing of markets, labor flows, and the possibility and financial viability of cross-country investments-are needed to encourage diversification. Are there tradeoffs, for example, between the desire of individual capital surplus countries to diversify holdings of assets and the concept of using surplus export income to support regional investment funds?

How to Approach Redistribution? Unlike normal economies, rents represent a high share of GDP in oil exporters. How these rents are "owned" and how they are distributed across the citizenry has to be a central issue, since this will affect incentives, especially on the labor market. If governments are forced to make transfers to their population in one

way or another, this implies an important divergence between private and social costs of employment. Wage payments will be a cost to private employers. But from the perspective of an oil-exporting government, a wage in an industrial firm might not be a cost but a benefit, relative to the alternative of making a transfer in another way (public employment, subsidy, etc) if it provides opportunities for productive activity and gain in skills. Seen in this light, policies that encourage and subsidize private employment could be seen as a form of conditional transfer program.

7.1 A Three-Policy Country Framework

Theory and country experience outside MENA suggest that three policies will be important for diversification.

1) Limiting booms and busts and managing the exchange rate to avoid extreme overvaluation that can force the economy into premature specialization on resources and non-traded goods (Hausmann and Velasco 2005). As well as general fiscal and monetary policies, measures can include selective policies, for example, to limit the flow of resources into key, non-traded sectors (real estate) during booms. Such policies have been used by some oil exporters (Malaysia).

2) A reasonably open trade policy, to limit the severe overvaluation of economies that will, by their nature, have high import coefficients.

3) More or less selective measures to encourage investment in non-oil tradable sectors and to bring down the costs of production in these sectors. They can include broader measures such as human capital formation, general infrastructure, and overall business climate reform, as well as more targeted measures directed to certain sectors or activities. The latter can include tax concessions and subsidies, specific infrastructure, industrial parks, etc., they can also be tailored to encouraging exports or to production for the domestic market.

Successful resource-based countries such as Malaysia, Chile, or Indonesia have followed variants of such policies (Gelb and Grasmann, 2010). But there may be some constraints, due to location or environment, that severely limit what particular countries can do in particular sectors. Despite good management and generous industrial incentives, sparse and relatively remote Botswana has not been successful in establishing a robust industrial sector to diversify away from diamonds. Within the diamond-processing industry also, given the very low transport cost of the raw commodity, its industrial activities are not competitive with the leading global complex in India which has access to superior technology, scale economies, and higher-productivity (and possibly cheaper) labor.

7.2 Microeconomic and Institutional Research Issues

Research in this area could start off from the body of work on industrial policy broadly defined. Harrison and Rodriguez-Clare (2010) provide an extensive review of research in this area and its conclusions. Other studies, such as Chandra (2006) provide detailed case material on the role of policies in encouraging export diversification in countries such as Chile, Malaysia, and Kenya. For the MENA region, Galal (2008) and contributors analyze the use of industrial policy in several countries, including case studies on Egypt, Morocco, Turkey, and Jordan. Other contributors review the motivation for and experience of industrial policies in Asia and the implications for the Middle East, and also consider the political economy of industrial policy.

From this body of research, it is clear that there is no simple answer to the question of whether industrial policy, in the "vertical" sense of targeted incentives directed towards a particular set of industries, has been effective or not. Noland and Pack (2008) suggest that the key to East Asian success has been far more due to "horizontal" policies that benefit many sectors, though not necessarily equally. These measures included encouraging rapid and high-quality human capital formation, an area where the Asian economies have been outstanding, and sustained high savings. This pattern is contrasted with that for the MENA region, including by Nabli et al (2008) who note the dominance of "vertical" industrial policy, often involving state-led industrialization and the dominance of sector control and decision-making of relatively "closed" groups, often with strong ties to the state and supported by oil-funded spending. The result has been relatively inward-looking production systems, lacking technological dynamism and unable to compete successfully on world markets. In contrast, despite a generally gloomy take on active industrial policy, Noland and Pack find that the one area where the Asians clearly got it right was in calibrating assistance to export performance, since this provided perhaps the one clear indicator of progress in competitiveness.

Studies of the Asian experience also flag the institutional structure and mechanisms of engagement between government and private sectors:the use of targets, incentives and recognition (especially for export success), and the ruthlessness with which the Asians were prepared to terminate some preferential measures to firms that failed to achieve export. The studies also show committee structures set up to jointly review evidence of progress, as an insurance against capture. This contrasts with the picture of sustained, long-term relationships between the state and industrial leaders in MENA, the lack of transparency in the region, and the development of interest groups that have perpetuated the use of vertical policies long past their usefulness. The result has been stagnant, uncompetitive industries less able to make an economic contribution to growth and employment creation.

The objective of research would therefore be to apply this body of work to the oil exporting countries in MENA. A first step could be to better understand how the MENA countries fit into global patterns in areas related to comparative advantage and competitiveness. A second component could be to go more deeply into the incentives in these countries, relating to prices, wages, and other elements of the business climate. A third strand could be case studies of diversification, or diversification potential, of selected MENA countries relative to comparators, possibly on a sector basis.

7.3 Application of Product Space Concepts

Recent research on product space (see section 2.3) may be used to get a more refined picture of the comparative situation of the oil exporters. This could include work to update and extend existing data bases on the sophistication and "closeness" of products and country export bundles. These would be used to benchmark the oil exporting countries, to provide an indication of the non-oil export areas that countries with their characteristics might be expected to consider, taking into account features such as their size, level of technical sophistication, and level of income. Given that the

current product mix is heavily weighted towards oil, does this approach provide useful indications of potential future comparative advantage for the oil exporters in other areas?

The research on product proximity and structure of product space could potentially have profound implications for the way we think about structural transformation and development policy and institutions. For example, two major contributors to this literature (Hausmann and Klinger, 2007) suggest a roadmap for future research, which includes the following research questions:

- First, does the structure of the product space matter for growth: specifically, do countries that specialize in the deeper (and denser) part of the product space grow faster?
- Second, what is the impact of distance to the frontier in a given product to the probability of upgrading to a new product
- Third, what factors affect the ability of countries to move to distant products: for example, from oil to biotechnology
- Fourth, is there a role for economic policy: is there a case for policies that could move a country from a sparse part of the product space to a denser one, then leave the rest to the natural process of proximity between products
- Fifth, do successful transitions, as in East Asia, usually happen as a consequence of countries being in the right part of the product space from the start or, instead, due to strategic move toward the denser part of the product space?

It would be important to extend these data and the approach to include international services, since these are likely to play an important role in the region.

7.4 Prices, Factor Incentives, and Market Failures

Market and coordination failures exist in all countries and are more pervasive in developing ones. Industrial policies may be advocated to compensate for them, including those preventing agglomeration externalities. For oil exporters in particular, industrial policies are often urged to contribute to another objective – to compensate for an "overvalued" exchange rate held at appreciated levels because of the exports of oil. But this may not be empirically evident in macroeconomic comparisons. In the 2005 ICP round, the US price level, with a PPP income of \$41,674 is 100. The unweighted PPP income average for Bahrain, Kuwait, Qatar, and Saudi Arabia is almost the same at \$40,547 but their average price level is only 69.5. Similarly Syria and Egypt, with PPP income a multiple of India or Bangladesh, have comparably low price levels, far below that of Tunisia or Morocco which have been more successful in export diversification.⁵⁶

While the MENA oil exporters may experience real exchange rate volatility depending on export prices, they are therefore not high-cost countries in the normal sense of having high prices relative to the Balassa norm for countries at their income level (Rodrik 2006b, 2007). No doubt this is because of pervasive subsidies on energy, food, and other goods, and low non-oil taxes; on the basis of comparative costs their real exchange rates are not appreciated. This may not, however, mean that they are low-cost in terms of production. The exchange-rate based measures do not capture regulatory costs or labor costs (in particular of nationals) relative to the productivity of the non-oil economy.

An interesting area for research could therefore be to analyze the cost elements of MENA's oil exporters in a comparative context, to understand how policies, including those on the distribution of oil rents, actually influence the cost structures of these countries relative to comparators. The approach could embody an element of value chain analysis, to benchmark particular sectors in the oil exporters with comparable industries in competitive countries.

What oil-related cost disadvantage, if any, would industrial policy aim to compensate for? If the issue is high productivity-adjusted labor costs or low levels of skill relative to that needed at the income level of the country, policy could include training grants, perhaps some employment subsidies, conditional on exports. If the issue is more one of density and first-mover costs, the approach might emphasize industrial zones.

7.5 Comparative Case Studies

Drawing on the matching process, research could include a number of comparative case studies of particular countries in MENA, including one or more comparators for each. The case studies would build on a common framework, to include: A) Overview:

- existing economic structure and endowments (including human capital, both national and diaspora)
- dependence on oil and how this shapes the economy
- (for comparators, a review of the pre-diversification situation)
- the imperative for diversification and the perceived nature of the inhibiting constraints

B) "Horizontal" Policies:

- macroeconomic management: expenditure stability and exchange rate management to provide a stable production base
- policies to build high-quality human capital; return and immigration policies;
- policies to reduce general costs of starting and growing businesses, including encouragement of new entrants, general tax, trade and investment policies

C) "Vertical" Policies that influence resource allocation, between (a) non-oil traded and nontraded sectors, and (b) between particular industrial or other traded sectors.

- selectivity in trade, investment, tax and public expenditure policies,
- other specific industrial policies, including those that discriminate between incumbents and new entrants

D) Institutional arrangements for implementing diversification policies

- the nature of the private sector and relationships with key government ministries
- the political economy of policy implementation: control or capture?
- the potential for institutional innovation for example, industrial zones are often urged to create islands of efficient infrastructure; perhaps they could be thought of also as potential islands of accountability and transparency?

7.6 Choice of Comparators

East Asian countries usually feature prominently in any discussion of diversification policy and economic success. They certainly can do so for the present research, which can build on, for example, the comparison between MENA countries by Nolan and Pack. But there are several other types of comparisons that might be interesting, which could also be used to encourage joint work by researchers in the region and outside:

One special category of comparators could be the group of hydrocarbon exporters, including some countries outside the region, such as Trinidad and Tobago and Indonesia, which have made special efforts to diversify within the hydrocarbon sector, including towards sectors such as petrochemicals, fertilizers, steel. There are several notable examples in MENA, including Saudi Arabia and Algeria.

Another special comparator set could be small states, such as Singapore and Mauritius, that have made and are still making particular efforts to diversify their economies, whether towards industrial or service-based and knowledge-based activities. These might be of special interest to the GCC countries.

A third set of comparators could be (country x sector) cases of relevance to MENA. For example, the experience of Central American countries in moving to computer assembly and high-level tourism (Costa Rica), diversification in Guatemala, and export-processing zones in the Dominican Republic. Why, for example, should Syria or Egypt not become the assembly point for electronics exports to the EU?

7.7 Data

A first component of research could involve stocktaking of the major data bases, both globally and for MENA countries, relating to a number of variables: country macroeconomic and sector data governance and business climate, including issues relating to regulation, firm performance, and economic/trade structure. These indices could include: WDI macroeconomic and sector data, human resources, governance databases, business climate and firm surveys, labor market and comparative costs, UNIDO industrial data bases and trade data and sophistication and logistics indices. The aim of this research would be to better understand where the MENA countries fit into the global picture and to locate possible comparators for the oil exporters, both locally and outside the region, considering especially the two groups of countries. What do these cases suggest about the potential for diversification, into broad categories of industrial, service, and other activities?

Conclusions

This report highlights the important role of oil in Arab economies, and discusses the promises and challenges it has entailed for these economies. In particular, the report analyzed the evidence on the tendency of oil economies, including in the Arab world, to experience frequent post-boom growth deceleration and long-term stagnation, and for some cases absolute decline of output--in other words an "oil curse". With the objective of identifying the most relevant issues for developing a strategy for escaping the oil curse, this report also undertakes a selective review of the literature on the oil curse.

The mandate of this approach report was to propose frontier areas for future research that are also of high policy relevance. In this context, the report proposes a potential future research agenda on two broad sets of themes. The first is on the macro-institutional issues for escaping the oil; and consequently on the strategies for escaping it; or, in other words, for harnessing the oil resource for long-term development of the Arab world. The second theme is on the micro-institutional issues of economic diversification, where the report addresses some pivotal questions, such as why and how to diversify, and for what development objective.

Notes

- 1. Elbadawi and Makdisi (2007) identify oil and conflicts as two factors explaining the "democracy deficit" in the Arab world and show that controlling for these two factors eliminates the significant Arab dummy that shows in modernity models of democracy. Arguably, at least some conflicts in the region might, in turn, be endogenous to oil.
- 2. The reserves estimates in the IMF (2009) report include Iran and are equal to 65% for oil and 45% percent for natural gas.
- 3. The Palestinian Territories, which belong to this group, are excluded for lack of data.
- 4. Comoros and Somalia, which belong to this group, are excluded for lack of data.
- 5. This happens until countries attain the income level of Ireland. Therefore, growth is associated with a high degree of specialization only at relatively high levels of income.
- 6. See also Klinger and Lederman (2004) and Carrere, Strauss-Kahn and Cadot (2007), who confirm Imbs and Wacziarg's pioneering work using more recent data sets.
- 7. The empirical literature uses a variety of measures to capture export diversification. Elbadawi (2002) uses a measure that is the residual of exports after the ten largest threedigit commodity groups have been accounted for. Imbs and Wacziarg (2003) capture concentration (the inverse of diversification) through the use of a Herfindahl-Hirschman index (HHI), coefficients of variation of sector shares, and maximum-minimum spreads. Lederman and Maloney (2007) also make use of HHI as well as of the share of natural resources in total exports.
- 8. The share of oil in Sudan's exports has exceeded 90% since 2006; and proceeds from oil also accounts for more than 70% of the total public sector revenues.
- 9. However, the share manufactured to total exports for SSA reported in this table is much larger than other estimates, including those of the IMF (2007).
- 10. This subsection draws heavily from Elbadawi, Kaltani, and Soto (2009).
- 11. Tunisia, for example, has been frequently referred to as an example of successful industrial transformation in the Arab region.

- 12. They estimate system GMM dynamic panel regressions, covering 96 countries during 1993-2004, which is the only period for which the EXPY data is available.
- 13. As in the received literature these include factors of production such as population, land per worker, natural resources, and geographic variables.
- 14. According to the definition of the RER adopted here, an RER overvaluation (undervaluation) is associated with a higher (lower) level of RER relative to its equilibrium.
- 15. Imbs and Wacziarg (2003) indicate that poor countries tend to diversify their production structure, but beyond a certain income threshold, further growth is associated with product concentration. However, Elbadawi et al regressions do not concentrate on this effect, because their sample is confined to developing countries.
- 16. See Rodrik (2007) for the RER estimation for China, Indonesia, and Malaysia among others, and Elbadawi et al (2009) for the case of Sudan.
- 17. More recent contributions have questioned Rodrik's earlier assessment of the sophistication of Chinese exports, arguing that they are in line with China's level of development (e.g. Kumakura, 2007).
- 18. For example, China was a subject of criticism by some US lawmakers because it has pursued a policy of strategic RER undervaluation to "unfairly" increase its competitiveness in the American market.
- 19. For other work in the literature on the role of RER undervaluation in promoting growth and export diversification, see Williamson (1997) and Elbadawi and Helleiner (2004).
- 20. Country examples from Rodrik (2007) are China, the Republic of Korea, Taiwan, Uganda, and Tanzania.
- 21. Notice that RCA is essentially equal to the weight of the PRODY index (equation 2.2).
- 22. Hidalgo et al (2007) find that the product space is indeed very sparse as suggested by the distribution of ϕ ; with 5% of its elements equal to zero, 32% of them smaller than 0.1 and 65% taking values below 0.2.
- 23. Quoted in Lederman and Maloney (2008).
- 24. This sub-section draws heavily from Elbadawi and Kaltani (2008).

- 25. For example, using a computable general equilibrium model with a 20-year horizon, Gelb and Associates (1988) estimate optimum savings rates as high 60–70% of the oil revenue boom; and Elbadawi and Majd (1993) also derive similar estimates in an inter-temporal two assets (oil and non-oil) welfare maximizing model applied to the oil surplus economies of Gulf Cooperation Council (GCC). These optimum savings ratios are much higher than those achieved by the oil exporting countries during boom times.
- 26. This draws heavily from Heal (2007).
- 27. This shadow price is the value of an extra unit of the good, which equals the market price (if it exists) corrected for external costs or benefits associated with the use of the good.
- 28. This is one of three cases analyzed by Heal as an extension to the Hotelling's model (1931).
- 29. For example, in any solution to this problem the social returns for both types of capital must be equal if there is investment in both (i.e. $u' = \lambda_f = \lambda_d$ if both I_f and I_d are positive). At the optimum, however, the shadow prices should change at different rates, since they reflect different fundamentals (the change in λ_f should reflect the difference between the discount rate and the interest rate on foreign assets, while the change in λ_d should reflect the difference between the discount rate and the efficiency in the extraction technology: α). Hence due to the contradictory implications of these conditions a country cannot spend on the two types of capital at the same time. The likely sequence is that countries would invest first to build their extraction capital (or domestic capital in general) until an appropriate level of domestic capital is built before switching to invest overseas, leaving extraction (or domestic capital) constant.
- 30. The formal empirical definition of "genuine savings" is public and private savings at home and abroad (net of depreciation) plus current spending on education to capture intangible human capital minus depletion of natural exhaustible and renewable resources minus the damage due to the stock of pollutants (co, and particulate).
- 31. See also Heal (2007), who further adjusted the World Bank's calculations for productivity and population growth. However, the quali-

tative conclusion remains unaffected.

- 32. For example, Collier et al (2009) note that because different forms of investment might have different rates of return, the composition as well as level of savings matter. In this context, they point out that if the domestic rate of return is double the world interest rate (at which, according to the Hotelling (1931) rule, the rent on the resource is expected to increase), the depletion of \$1m of natural assets would be fully offset by \$0.5 m of domestic investment.
- 33. This follows very closely the exposition of Sachs (2007).
- 34. In this simple graphical model, the expression for the RER that is consistent to non-traded goods equilibrium is given by a linear equation: $E_N = \text{constant-} 1/(p_N/p_T) E_T = \text{constant-} (1/\text{RER})E_T$.
- 35. See for example Hnatkovska and Loayza (2004), Inter-American Development Bank (1995), Gavin and Hausmann (1996), Ramey and Ramey (1995), Aizenman and Marion (1999), Caballero (2000), Flug et al. (1999), and Duryea (1998).
- 36. Hamilton (2008) demonstrates that the random walk standard error for an "s" quarter forecast into the future becomes $\sigma \sqrt{s}$.
- 37. However, as Gelb and Grasmann suggest there are also institutional and political factors, associated with the asymmetric nature of the political payoff, that tend to limit the use of hedging through future contracts, forward markets, commodity swaps, and bonds. In particular, politicians get little visibility and political benefits for profits made, but losses open them to charges of misuse of public money and corruption.
- 38. See for example, Cover (1992) for the USU.S., and Kandil (1999) for the Middle East and the more recent paper by Collier and Venables (2008), who provide a summary of research on the impact of terms of trade shocks in developing countries.
- 39. In their book on escaping the resource curse, Humphreys et al (2007) devote considerable attention to these issues. See also Collier (2008) and Collier and Goderis (2009), who strongly advocate auctioning as an instrument for selling oil rights by countries emerging out of conflicts. These countries tend to

lack capable institutions and knowledge and are highly susceptible to corruption.

- 40. Gelb and Grasmann (2008) suggest that the markets for oil exploration and production are experiencing a rapid change with the recent emergence of state owned oil companies from relatively advanced energy importing developing countries, such as China, India, and Malaysia. This, they argue, has exerted a lot of competition that is bound to favour emerging small oil producers. Also, Elbadawi and Kaltani (2008) made a similar assessment of the increasingly active role of China on sub-Saharan Africa's energy sector.
- 41. However, there two exceptions among the oil-rich group: Kuwait, which is ruled by what can be characterized as the closest Arab regime to a constitutional monarchy; and Algeria, which is ruled by a partial democracy (Elbadawi and Makdisi, 2010).
- 42. For an analysis of the common-pool problem see, for example, von Hagen and Harden (1994).
- 43. See, for example, Hausmann, Powell, and Rigobon (1993); and Lane and Tornell (1999).
- 44. Social conflicts and lack of effective institutions for mediating them could affect the response to external shocks in many ways: by delaying adjustment in fiscal policy and in key relative prices, most notably the real exchange rate; by generating increased uncertainty in the economic environment; and by diverting resources from productive to distributive activities.
- 45. They built a new series of resource rents using the following methodology. First, they defined rents as the difference between natural resource price and extraction costs. Second, for each point-source natural resource, they multiplied the estimated rents per unit of output by the volume extracted; then they add up the outcome for such resources (e.g. oil, gas, coal, iron, copper, etc). Finally, for each year they divide total rents by GDP for the country in question.
- 46. See also Ossowski et al (2008), who in a recent IMF Occasional Paper test the effectiveness of fiscal institutions in explaining non-oil primary balance. However, arguably this is not very informative on the issue of countercyclicality (for a critique of the IMF's perspec-

tive on this issue, see Collier et al, 2009).

- 47. These include pensions; minimum revenue guarantees issued to private infrastructure concessions; government guarantees on bank deposits and quasi-fiscal losses of the Central Bank (Schmidt-Hebbel, 2007).
- 48. Quoted from Eifert et al. (2003), who comment that although Karl's diagnosis may seem overly deterministic, it nevertheless offers a fairly accurate description of how oil rents are managed in many countries.
- 49. The following paragraphs draw heavily from Humphreys and Sandbu's article.
- 50. See for example Lederman and Maloney (2007), who argue that, unlike the resource dependency measure, the abundance measure is consistent with theory; and that when used as a proxy for resource rents no oil curse is found. Instead, they find a robust positive association between resource wealth and long-term growth.
- 51. See, for example, Arezki and van der Ploeg (2008).
- 52. The PMG is developed by Pesaran, Shin, and Smith (1999).
- 53. These are briefly summarized in section 5; however, for a detailed discussion see Humphreys and Sandbu (2007).
- 54. Razzak (2007) and Elbadawi and Kamar (2005) are among the few exceptions.
- 55. See Williamson (2005).
- 56. The PPP price levels for particular components of absorption, such as food, may of course be different from the overall ICP price level, an area for research. However, subsidies may be one reason why dummy variables for oil exporter status are significant in explaining lack of diversification even when exchange rate variables are included; the measures of the latter do not fully account for the incentives working against other sectors.

References

- Acemoglu, D., S. Johnson and J.A. Robinson. 2001.
 "The Colonial Origins of Comparative Development: An Empirical Investigation." *American Economic Review* 91, 5: 1369-1401.
- Aghion, P. & P. Howitt. 1992. "A Model of Growth Through Creative Destruction." *Econometrica*

6(2): 323-351.

- Aizenman, Joshua and Nancy Marion. 1999. "Volatility and Investment: Interpreting Evidence from Developing Countries." *Economica*, 66: 157–79.
- Arezki, R. and F. Van der Ploeg. 2008. "Can the Natural Resource Curse be Turned into a Blessing? Role of Trade Policies and Institutions." IMF Working Paper 07/55, International Monetary Fund, Washington, DC.
- Assaad, R. 2002. "The Transformation of the Egyptian labor market: 1988-98." In *The Egyptian Labor Market in an Era of Reform*. Edited by R. Assaad. Cairo: American University in Cairo Press.
- Auty, Richard. 1990. *Resource-based Industrializtion: Sowing the Oil in Eight Developing Countries.* Clarendon Press, Oxford, UK..
- Bigsten, Arne, Paul Collier, Stefan Dercon, Marcel Fafchamps, Bernard Gauthier, Jan Willem Gunning, Abena Oduro, Remco Oostendorp, Catherine Pattillo, Mans Soderbom, Francis Teal and Albert Zeufack. 2002. "Do African Manufacturing Firms Learn from Exporting?" Working Paper WPS/2002-09. Centre for the Study of African Economies, Oxford, UK.
- Budina Nina and Sweder van Wijnbergen. 2008. "Managing oil revenue volatility in Nigeria: The role of fiscal policy". Chapter 10 in *Africa at Turning Point*? Delfin and Page (eds.), World Bank.
- Caballero, Ricardo. 2000. "Macroeconomic Volatility in Latin America: A View and Three Case Studies." *Economia*. 1(1): 31-108.
- Caballero, Ricardo. 2002. "Coping with Chile's External Vulnerability: A Financial Problem." Working Papers Central Bank of Chile 154, Central Bank of Chile.
- Caballero, Ricardo and Arvind Krishnamurthy. 2000. "International and Domestic Collateral Constraints in a Model of Emerging Market Crises." NBER Working Papers 7971, National Bureau of Economic Research, Inc.
- Calderón, C. and K. Schmidt-Hebbel. 2003. "Macroeconomic Policies and Performance in Latin America." *Journal of International Money and Finance* 22, 895-923.
- Carrere, C.V. Strauss-Kahn and O. Cadot. 2007. "Export Diversification: What's Behind the Hump?" Clermont-Ferrand, Centre d'etudes et de recherché sur le development international

(CERDI, Etudes et documents, E 2007.24).

- Chandra, V. 2006. *Technology, Adaptation and Exports How some Developing Countries Did It,* The World Bank.
- Collier, Paul, Frederick van der Ploeg, Michael Apence and Anthony J. Venables. 2009. "Managing Resource Revenues in Developing Economies." OxCarre Research Paper 15, The University of Oxford.
- Collier, Paul and Venables, Anthony J., 2008. "Illusory revenues: Tariffs in resource-rich and aid-rich economies." CEPR Discussion Papers, 6729, C.E.P.R. Discussion Papers.
- Collier, P. and B. Goderis.2009. "Structural policies for shock-prone developing countries." CSAE Working Paper No. 2009-03, Department of Economics, University of Oxford.
- Collier, P. and B. Goderis. 2008. "Community prices, growth and the natural resource curse: Reconciling a conundrum," mimeo, Centre for the Study of African Economies, University of Oxford.
- Collier, P. and A. Hoeffler. 2009. "Testing the Neocon Agenda: Democracy and Natural Resource rents." *European Economic Review*, 53(3): 293-308.
- Collier, P. and A. Hoeffler. 2004. "Greed and Grievance." Oxford Economic Paper, 56(4): 563-595.
- Corden, W.M. 1982. "Exchange Rate Policy and the Resource Boom." *Economic Record*, 58(160): 18–31.
- Corden, W.M. and J.P. Neary. 1984. "Booming Sector and De-industrialization in a Small Open Economy." *Economic Journal*, 92: 825–48.
- Cover, James P. 1992. "Asymmetric Effects of Positive and Negative Money-Supply Shocks." *The Quarterly Journal of Economics*, 107(4): 1261-1282.
- Cuddington, John T., Rodney Ludema, and Shamila A. Jayasuriya. 2007. "Prebisch-Singer Redux." In W. Maloney and D. Lederman, eds., Natural Resources: Neither Curse nor Destiny, pp. 15–40. Stanford, California: Stanford University Press for The World Bank.
- Davis, Jeffrey M. Rolando Ossowski and Anna Fedelino (eds). 2003. *Fiscal Policy Formulation and Implementation in Oil Producing Countries*. IMF, Washington, DC.
- Duryea, Suzanne. 1998. "Children's Advancement through School in Brazil: The Role of Transitory Shocks to Household Income." Inter-American

Development Bank WPS-376.

- Easterly, W. and R. Levine. 2003. "Tropics, Germs, and Crops: How Endowments Influence Economic Development." *Journal of Monetary Economics*, 50: 3-39.
- Eichengreen, Barry. 2007. "The Real Exchange Rate and Economic Growth." Paper prepared for the World Bank's Commission on Growth.
- Eifert, Benn, Alan Gelb, and Nils Borje Tallroth. 2003. "The Political Economy of Fiscal Policy and Economic Management in Oil Exporting Countries." in Davis, J.M. et al (eds.)
- Elbadawi, Ibrahim. 2002. "Real Exchange Rate Policy and Non-traditional Exports in Developing Countries." In G.K. Helleiner, ed., Non-Traditional Export Promotion in Africa: Experiences and Issues. New York: Palgrave.
- Elbadawi, Ibrahim. 2005a. "Reviving Growth in the Arab World." *Economic Development and Cultural Change*, 53: 293–26.
- Elbadawi, Ibrahim. 2005b."The Politics of Sustaining Growth in the Arab World: Getting Democracy Right." Working Paper Series No. 2, 2004, American University of Beirut. Institute of Financial Economics.
- Elbadawi, Ibrahim. 2005c. "Institutions, Trade and Geography: Implications for Long-term development in the Arab World." *Journal of Development and Economic Policy*, 8(1): 7-26 (December).
- Elbadawi, Ibrahim and Samir Makdisi (eds). 2010. Democracy in the Arab World: Explaining the Deficit. Routledge, Routledge Studies in Middle Eastern Politics, Ottawa, Cairo.
- Elbadawi, Ibrahim, Samir Makdisi and Gary Milante. 2010. "Explaining the Arab Democracy Deficit: The Role of Oil and Conflicts." In Elbadawi and Makdisi (2010) *Democracy in the Arab World: Explaining the Deficit*. Routledge, Routledge Studies in Middle Eastern Politics, Ottawa, Cairo, p. 41-82.
- Elbadawi, Ibrahim, Linda Kaltani, and Raimundo Soto. 2009. "Aid, Real Exchange Rate Misalignment And Economic Performance in Sub-Saharan Africa." Unpublished mimeo, The World Bank, Washington DC.
- Elbadawi, Ibrahim and Raimundo Soto. 2009. "Theory and Empirics of Real Exchange Rates in Developing Countries." Unpublished mimeo, The World Bank, Washington DC.

Elbadawi, Ibrahim and Norman Loayza. 2008.

"Informality, Employment and Economic Development in the Arab World." Journal of Development and Economic Policies; 2(2): 27-75.

- Elbadawi, Ibrahim, Linda Kaltani, and Klaus Schmidt-Hebbel. 2008. "Foreign Aid, the Real Exchange Rate, and Growth in the Aftermath of Civil Wars." *World Bank Economic Review*, 22(1), March.
- Elbadawi, Ibrahim and Linda Kaltani. 2008. "The Macroeconomics of Oil Booms: Lessons for Sub-Saharan Africa" in Matthew Martin (editor). *Managing Commodity Booms in Sub-Saharan Africa*. The African Economic Research Consortium, Nairobi, Kenya.
- Elbadawi, Ibrahim and Samir Makdisi. 2007. "Democracy and development in the Arab World." in Elbadawi, I., H. Esfahani and H. Kheir-El-Din (editors): Special Issue of the Quarterly Review of Economics and Finance: 46(5): 813-831
- Elbadawi, Ibrahim and Bassem Kamar. 2005. "The Great Debate on Exchange Rate Regimes: Why Should the MENA Region Care?" Unpublished mimeo, The World Bank, Washington DC.
- Elbadawi, I. and G. Helleiner (2004). "African Development in the Context of the New World Trade and Financial Regimes: The Role of the WTO and its Relationship to the World Bank and IMF," in A. Oyejide and W. Lyakurwa (editors). *Africa and the World Trading System*, Volume I: Framework Paper. MacMillan: chapter 9.
- Elbadawi, Ibrahim and Nader Majd. 1993. "Managing a non-renewable resource: Savings and exchange rate policies in Bahrain." Policy Research Working Paper No. 1134. Policy Research Department, The World Bank, Washington, D.C., April.
- ERF. 1998. *Economic Trends in the MENA Region*. Economic Research Forum, Cairo.
- Fasano, Ugo. 2000. "Review of the Experience with Oil Stabilization and Savings Funds in Selected Countries" IMF Working paper # WP/00/112, Washington DC.
- Flatters, Frank. 2005. "The Economics of MIDP and the South African Motor Industry." Prepared for presentation and discussion at TIPS/ NEDLAC South Africa Trade and Poverty Programme (SATPP) Policy Dialogue Workshop, Johannesburg, 2 November (available at www. frankflatters.com).

Flug, Karnit, Antonio Spilimbergo and Eric Wa-

chtenheim. 1999. "Investment in education: Do economic volatility and credit constraints matter?" *Journal of Development Economics*, 55(2): 465-481.

- Frankel, Jeffrey and Ayako Saiki. 2002. "A Proposal to Anchor Monetary Policy by the Price of the Export Commodity." *Journal of Economic Integration* 17(3) (September): 417-448.
- Galal, Ahmed. 2008. *Industrial Policy in the Middle East and North Africa: Rethinking the Role of the State.* The Egyptian Center for Economic Studies Publication.
- Galal, Ahmed. 2002. "The Paradox of Education and Unemployment in Egypt." Working Paper No. 67. The Egyptian Center for Economic Studies. Cairo, Egypt.
- García, M., P. García and B. Piedrabuena. 2005. "Fiscal and Monetary Policy Rules: the Recent Chilean Experience." Working Paper 340, Central Bank of Chile.
- Gavin, Michael and Ricardo Hausmann. 1996. "Securing Stability and Growth in a Shock Prone Region: The Policy Challenge for Latin America." WPS-315. Inter-American Development Bank.
- Gelb, Alan and Associates. 1988. *Oil Windfalls: Blessing or Curse? World Bank*. Oxford University Press.
- Gelb, Alan and Sina Grasmann. 2010. "How Should Oil Exporters Spend Their Rents?" Working Papers 221, Center for Global Development, Washington DC.
- Gelb, Alan and Sina Grasmann. 2008. "Confronting the Oil Curse." Unpublished mimeo, The World Bank, Washington DC.
- Grossman, G. & E. Helpman. 1989. "Product development and international trade." *The Journal of Political Economy* 97(6): 1261 – 1283.
- Hamilton, James D. 2008. *Understanding Crude Oil Prices*. University of San Diego, Draft June 4.
- Harrison, AE and Andres Rodriguez-Clare. 2010. "Trade, Foreign Investment, and Industrial Policy," Handbook of Development Economics.
- Hausmann Ricardo. 2001. "Venezuela's Growth Implosion: A Neo-classical Story?" In Dani Rodrik (editor), In Search of Prosperity: Analytic Narratives on Economic Growth, Princeton University Press.
- Hausmann, Ricardo and Baily Klinger. 2007. "The Structure of the Product Space and the Evolu-

tion of Comparative Advantage." Center for International Development Working Paper no. 146, (April).

- Hausmann, Ricardo, Jason Hwang and Dani Rodrik. 2006. "What you export matters." Available at http://ksghome.harvard. edu/~drodrik/hhr.pdf.
- Hausmann, Ricardo and Dani Rodrik 2006. "Doomed to Choose: Industrial Policy as Predicament." Unpublished mimeo, Kennedy School of Government, Harvard University.
- Hausmann, Ricardo, Lant Prichett, and Dani Rodrick. 2005. "Growth Accelerations." *Journal of Economic Growth*, 10(4): 303-329.
- Hausmann Ricardo, and Andres Velasco. 2005. " The Causes of Financial Crises: Moral Failure Versus Market Failure" in M.A. Centano, H. James, and J. Londregan (editors). *The Political Economy of Recurrent Debt*. Princeton Institute for International and Regional Studies.
- Hausmann, Ricardo and Roberto Rigobon. 2003. "An alternative interpretation of the 'Resource Curse: theory and Policy Implications.'" in Davis et al (eds).
- Hausmann, Ricardo, Andrew Powell and Roberto Rigobon. 1993. "An Optimal Spending Rule Facing Oil Income Uncertainty (Venezuela)." In E. Engel and P. Meller, eds., *External Shocks* and Stabilization Mechanisms, pp. 113–71. Baltimore: Johns Hopkins University Press for Inter-American Development Bank.
- Heal, Geoffrey. 2007. "Are Oil Producers Rich." In Humphreys, Macartan, Jeffrey D. Sachs and Joseph E. Stiglitz (eds.): p. 155-172.
- Hidalgo, C.A., B. Klinger, A.L. Barabasi and R. Hausmann. 2007. "The Product Space Conditions the Development of Nations." *Science*, 317: 482-487.
- Hotelling, H. 1931. "The Economics of Exhaustible Resources." *Journal of Political Economy*, 39(2): 137-75.
- Humphreys, Macartan, Jeffrey D. Sachs and Joseph E. Stiglitz (eds.). 2007. *Escaping the Resource Curse*. Columbia University Press, New York.
- Humphreys, Macartan and Martin E. Sandbu. 2007. "The Political Economy of Natural Resource Funds." In Humphreys, Macartan, Jeffrey D. Sachs and Joseph E. Stiglitz (eds.): p. 194-234.
- Hnatkovska, Victoria and Norman Loayza. 2004.

"Volatility and growth." Working Paper Series No. 3184. The World Bank, Washington, D.C.

- Imbs, Jean and Romain Wacziarg. 2003. "Stages of Diversification." American Economic Review, 93 (1), March: 63-86.
- IMF. 2009. Regional Economic Outlook: Middle East and Central Asia. The International Monetary Fund, Washington DC: October.
- Inter-American Development Bank. 1995. Economic and Social Progress in Latin America: 1995 Report.
- Johnson, Simon, Jonathan Ostry, and Arvind Subramanian. 2007. "Africa's Growth Prospects: Benchmarking the Constraints." IMF Working Paper WP/07/52, Washington: International Monetary Fund.
- Kandil, Magda E. 1999. "The Asymmetric Stabilizing Effects of Price Flexibility: Historical Evidence and Implications." *Applied Economics*, 31(7): 825-839.
- Karl, Terry. 1999. "The Perils of the Petro-state: Reflections on the Paradox of Plenty." *Journal of International Affairs (US)*, 53(1): 31–48.
- Klinger, B. And D. Lederman. 2004. "Discovery and Development: An Empirical Exploration of 'New' Products." (World Bank Policy Research Working Paper No. 3450).
- Korhonen, Iikka & Tuuli Juurikkala. 2007. "Equilibrium exchange rates in oil-dependant countries." BOFIT Discussion Papers 8/2007, Bank of Finland, Institute for Economies in Transition.
- Kumakura, Masanaga, 2007, "What's so special about China's exports? A comment." China & World Economy, 15(5): 18-37.
- Kumhof, Michael and Douglas Laxton. 2009.
 "Chile's Structural Fiscal Surplus Rule: A Model-Based Evaluation." IMF Working Papers 09/88: International Monetary Fund, Washington DC
- Lane, Philip and Aaron Tornell. 1999. "The Voracity Effect." American Economic Review, 89: 22–46.
- Leamer, Edward E. 1987. "Paths of Development in the Three Factor, N-good General Equilibrium Model." *The Journal of Political Economy* 95(5): 961 – 999.
- Leamer, Edward E. 1984. Sources of International Comparative Advantage: Theory and Evidence. Cambridge, Massachusetts, The MIT Press.

Leigh, D. And J.P. Olters 2006. "Natural resource

Depletion, Habit Formation, and Sustainable Fiscal Policy: Lessons from Gabon," IMF Working Paper 06/193, International Monetary Fund, Washington DC.

- Maloney, William and Daniel Lederman. 2008. "In Search of the Missing Resource Curse." Unpublished mimeo, The World Bank, Washington DC.
- Maloney, William and Daniel Lederman. 2007. "Trade Structure and Growth." In W. Maloney and D. Lederman, eds., *Natural Resources: Neither Curse nor Destiny*, pp. 15–40. Stanford, California: Stanford University Press for The World Bank.
- Manzano, Osmel and Roberto Rigobon (2007), "Resource Curse or Debt Overhang?" In W. Maloney and D. Lederman, eds., Natural Resources: Neither Curse nor Destiny, pp. 41–70. Stanford, California: Stanford University Press for The World Bank.
- Matsen, Egil and Ragnar Torvik. 2005. "Optimal Dutch Disease." *Journal of Development Economics*, 78(2): 494-515, December.
- Matsuyama, Kiminori. 1992. "Agricultural Productivity, Comparative Advantage, and Economic Growth." *Journal of Economic Theory*, 58(2): 317-334, December.
- Melhum, Halvor, Karl Moene, and Ragnar Torvik. 2005. "Institutions and the Resource Curse." *Economic Journal*, 116(508): 1-20.
- Mongardini, Joannes, and Brett Rayner. 2009. "Grants, Remittances and the Equilibrium. Real Exchange Rate in Sub-Saharan African Countries." IMF Working paper # WP/09/115, Washington DC.
- Nabli, Mustapha K. 2004. "Long term Economic Development Challenges and Prospects for the Arab Countries." Unpublished mimeo, The World Bank, Washington DC, February.
- Noland, Marcus and Howard Pack. 2008. "Arab Economies at a Tipping Point" *Middle East Policy*, 15(1): 60-69.
- Ossowski, R., M. Villlafuerte, P. A. Medas, and T. Thomas. 2008. "Managing the Oil Revenue Boom: The Role of Fiscal Institutions." IMF Occasional Paper # 260, Washington DC.
- Pesaran, H.; Y. Shin and R. Smith. 1999. "Pooled Mean Group Estimation of Dynamic Heterogeneous Panels." *Journal of the American Statistical Association*, 94:621-634.

Ramey, Garey & Ramey, Valerie A. 1995. "Cross-

Country Evidence on the Link between Volatility and Growth." *American Economic Review*, 85(5): 1138-51, December.

- Razzak, Weshah. 2007. "In The Middle of the Heat:The GCC countries Between Rising Oil Prices and the Sliding Greenback." MPRA Paper 6591, University Library of Munich, Germany.
- Reinhart, Carmen and Guillermo Calvo. 2000. "When Capital Inflows Come to a Sudden Stop: Consequences and Policy Options." MPRA Paper 6982, University Library of Munich, Germany.
- Rietveld, M., and R. Pringle. 2007. "The Evolution of Sovereign Wealth Management." In J. Johnson-Calari, and M. Rietveld, eds., *Sovereign Wealth Management*. London: Central Banking Publications Ltd.
- Robinson, J. A., R. Torvik and T. Verdier. 2006. "Political Foundations of the Resource Curse." *Journal of Development Economics*, 79: 447-468.
- Robinson, James A. and Torvik, Ragnar. 2005. "White Elephants." *Journal of Public Economics*, 89(2-3): 157-566.
- Rodrik, D. 2007. "Why Does the Real Exchange Rate Matter to Growth." Unpublished mimeo, Kennedy School of Government, Harvard University.
- Rodrik, Dani. 2006a. "Industrial Development: Stylized Facts and Policies." Unpublished mimeo, Kennedy School of Government, Harvard University.
- Rodrik, Dani. 2006b. "What's So Special about China's Exports?" Working Paper Series rwp06-001, Harvard University, John F. Kennedy School of Government, Harvard University.
- Rodrik, D., A. Subramanian and F. Trebbi. 2002. "Institutions Rule: The Primacy of Institutions over Geography and Integration in Economic Development." National Bureau of Economic Research Working Paper, November 2002.
- Rodrik, Dani. 1999. "Where Did all the Growth Go? External Shocks, Social Conflict and Growth Collapses." *Journal of Economic Growth*, 4(4): 385–412.
- Ross, Michael. 2003. "How Does Mineral Wealth Affect the Poor?" UCLA Department of Political Science, April.
- Sachs, Jeffrey and Andrew Warner. 1997. "Natural Resource Abundance and Economic Growth."

Updated version of 1995 NBER working paper, 5398. Center for International Development and Harvard Institute for International Development, November.

- Sachs, D. Jeffrey. 2007. "How to Handle the Macroeconomics of Oil Wealth." In Humphreys, Macartan, Jeffrey D. Sachs and Joseph E. Stiglitz (eds.): p. 173-193.
- Sachs, Jeffrey. 2003. "Institutions Don't Rule: Direct Effects of Geography on Per Capita income." National Bureau of Economic Research. Working Paper No. 9490.
- Sala-i-Martin, Xavier. 1997. "I just Ran Four Million Regressions," Economics Working Papers 201, Department of Economics and Business, Universitat Pompeu Fabra.
- Sala-i-Martin, Xavier, and Elsa v. Artadi. 2002. *Economic Growth and Investment in the Arab World*. World Economic Forum 2002. New York and Oxford: Oxford University Press.
- Schmidt-Hebbel, Klaus. 2007. "Fiscal policy in an open economy." Teaching Notes for the World Bank and the Sudanese Ministry of Finance Workshop on Analytical Tools for Macroeconomic and Fiscal Management, Khartoum, Sudan, January 2007.
- Sekkat, Khalid. 2009. "Does Competition Improve Productivity in Developing Countries?" Journal of Economic Policy Reform, 12(2): 145-162.
- Sekkat, K. and A. Varoudakis. 1998. "Exchange Rate Management and Manufactured Exports in Sub-Saharan Africa." Technical Paper No. 134. OECD Development Centre, Paris.
- Setser, Brad. 2007. "The Case for Exchange Rate Flexibility in Oil-Exporting Economies." Policy Brief No. PB07-8, Peterson Institute for International Economics: Washington DC.
- Smith, B. 2004. "Oil Wealth and Regime Survival in the Developing World, 1960-1999." *American Journal of Political Science* 48(2): 232-246.
- Stiglitz, Joseph. 2007. "What is the Role of the State?" In Humphreys, Macartan, Jeffrey D. Sachs and Joseph E. Stiglitz (eds.): p. 23-52.
- UNDP and Arab Fund for Economic and Social Development. 2002. Arab Human Development Report 2002: Creating Opportunities for Future Generations. New York: United Nations Development Program
- UNIDO. 2009. Breaking In and Moving Up: New Industrial Challenges for the Bottom Billion and the Middle-Income Group. Industrial De-

velopment Report 2009: United Nations, Industrial Development Organization.

- von Hagen, Weingast Jurgen and Ian Harden. 1994. "National Budget Processes and Fiscal Performance." *The European Economy, Reports and Studies* 3: 311–418.
- Williamson, John. 2005. "Exchange Rate Regimes for the MENA Region." Unpublished mimeo, The Peterson Institute for International Economics, Washington DC: December.
- Williamson, John. 1997. "Exchange Rate Policy and Development Strategy" in Elbadawi and Soto (eds.) Foreign Exchange Markets and Exchange Rate Policies in Sub-Saharan Africa, *Journal of African Economies*, 6(3): 17–36.
- World Bank. 2003. "MENA Development Report: Better Governance for Development in ME-NA—Enhancing Inclusiveness and Accountability." Washington, DC: World Bank.
- World Bank. 2004. Unlocking the employment potential in the Middle East and North Africa: Toward a New Social Contract. MENA Development Report, The World Bank, Washington, D.C.
- World Bank 2006. Where is the Wealth of Nations? Measuring Capital for the 21st Century. Washington DC.

About the Authors

Ibrahim Elbadawi is the Director of the Macroeconomics Research and Forecasting Department, Dubai Economic Council, and an ERF Research Fellow. Until recently he was Lead Economist at the Development Research Group of the World Bank, which he joined in 1989; and before that he was an Associate Professor of economics at the University of Gezira in Sudan. He has published widely on several areas of macroeconomic and development policy: exchange rate regimes, real exchange rate and macroeconomic competitiveness, macroeconomics of oil management and economic diversification; conflict, post-conflict transitions and peace-building; aid effectiveness and development partnership; and, investment climate and export performance.

Alan Gelb is is a senior fellow at the Center for Global Development. He had previously been Director of Development Policy at the World Bank; prior to this Chief Economist for the Africa Region and Staff Director for the 1996 World Development Report "From Plan to Market". His main recent areas of work have included the special development challenges of resource-rich countries, aid and development outcomes, the transition from planned to market economies, and Africa, including directing a major study "Can Africa Claim the 21st Century".